UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, DC 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): April 8, 2019 (April 2, 2019)



i3 Verticals, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or Other Jurisdiction of Incorporation) 001-38532 (Commission File Number) 82-4052852 (I.R.S. Employer Identification No.)

40 Burton Hills Blvd., Suite 415 Nashville, TN (Address of principal executive offices)

37215 (Zip Code)

(615) 465-4487 (Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d- 2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e- 4(c))

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company. X

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 1.01 Entry into a Material Definitive Agreement.

On April 3, 2019, i3 Verticals, Inc., a Delaware corporation (the "Company"), through i3 Verticals, LLC, a Delaware limited liability company and wholly owned subsidiary of the Company ("i3 LLC"), and i3-Bearcat, LLC, a Delaware limited liability company and wholly owned subsidiary of i3 LLC (the "Buyer"), completed its acquisition of Northeast Texas Data, LLC, a Texas limited liability company ("NET Data") and Graves Humphries Stahl, LLC, a Texas limited liability company ("GHS" and, together with NET Data, the "Public Sector Targets"), pursuant to the terms of a Membership Interest Purchase Agreement (the "Purchase Agreement"), dated April 3, 2019, by and among i3 LLC, Buyer, NTD Holdings, Inc., a Texas corporation ("NTD Holdings"), GH Holdco, Inc., a Texas corporation ("GH Holdco", and together with NTD Holdings, the "Sellers"), and David Graves and Tory Humphries, as owners of the Sellers. The Public Sector Targets are engaged in the business of developing, enabling, supporting, marketing, distributing and selling software products and services to small and midsize county governments and city court systems.

Pursuant to the terms of the Purchase Agreement, the Buyer acquired all of the issued and outstanding equity interests of the Public Sector Targets (the "Acquisition") in exchange for approximately \$24 million in cash, subject to certain pre- and post-closing net working capital adjustments (the "Acquisition Consideration"). In addition to the Acquisition Consideration, the Sellers are eligible to receive, subject to the satisfaction of certain conditions set forth in the Purchase Agreement related to the performance of the Public Sector Targets in the following 12 months, additional consideration of \$1 million. The Purchase Agreement contains customary representations, warranties, covenants and indemnities by the parties to such agreement.

The foregoing description of the Purchase Agreement does not purport to be complete and is qualified in its entirety by reference to the Purchase Agreement, a copy of which is filed herewith as <u>Exhibit 2.1</u> and incorporated herein by reference in this Item 1.01. The Purchase Agreement has been included as an exhibit hereto solely to provide investors and security holders with information regarding its terms. It is not intended to be a source of financial, business or operational information about the Company, i3 LLC, NTD Holdings, GH Holdco or their respective subsidiaries or affiliates. The representations, warranties and covenants contained in the Purchase Agreement are made only for purposes of the Purchase Agreement and are made as of specific dates; are solely for the benefit of the parties; may be subject to qualifications and limitations agreed upon by the parties in connection with negotiating the terms of the Purchase Agreement, including being qualified by confidential disclosures made for the purpose of allocating contractual risk between the parties rather than establishing matters as fact; and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors or security holders. Investors and security holders should not rely on the representations, warranties and covenants or any description thereof as characterizations of the actual state of facts or condition of the Company, i3 LLC, NTD Holdings, GH Holdco, the Public Sector Targets or their respective subsidiaries or affiliates. Moreover, information concerning the subject matter of the representations, warranties and covenants may change after the date of the Purchase Agreement, which subsequent information may or may not be fully reflected in public disclosures.

Item 2.01 Completion of Acquisition or Disposition of Assets.

See Item 1.01 of this Current Report on Form 8-K, which is incorporated herein by reference.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

On April 2, 2019, the Board of Directors (the "Board") of the Company, upon the recommendation of the Compensation Committee of the Board, approved an increase in the annual base salary for Greg Daily, the Company's Chief Executive Officer, to \$300,000 per year, effective April 1, 2019. Previously, Mr. Daily received only a nominal salary that was equivalent to the employee cost of his health insurance premiums (approximately \$12,000 per year).

Item 7.01 Regulation FD Disclosure.

On April 8, 2019, the Company issued a press release announcing the execution of the Purchase Agreement and the completion of the Acquisition. A copy of the press release is furnished as <u>Exhibit 99.1</u> hereto.

Item 9.01. Financial Statements and Exhibits.

(a) Financial statements of businesses acquired

The Company intends to file the financial statements required by Item 9.01(a), in accordance with Rule 3-05 of Regulation S-X, by amendment to this Current Report on Form 8-K no later than 71 calendar days following the date that this Current Report on Form 8-K is required to be filed.

(b) Pro forma financial information

The Company intends to file the pro forma financial information required by Item 9.01(b) by amendment to this Current Report on Form 8-K no later than 71 calendar days following the date that this Current Report on Form 8-K is required to be filed.

(d) Exhibits.

<u>Exhibit No.</u>	Description
	Membership Interest Purchase Agreement, dated April 3, 2019, by and among, i3 Verticals, LLC, i3-Bearcat, LLC, NTD Holdings, Inc., GH Holdco, Inc. and David Graves and Tory Humphries.
<u>99.1</u>	Press Release dated April 8, 2019

* Schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. i3 Verticals, Inc. hereby undertakes to furnish supplemental copies of any of the omitted schedules and exhibits upon request by the Securities and Exchange Commission.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: April 8, 2019

i3 VERTICALS, INC.

By: /s/ Clay Whitson

Name:Clay WhitsonTitle:Chief Financial Officer

MEMBERSHIP INTEREST PURCHASE AGREEMENT

BY AND AMONG

i3-BEARCAT, LLC

AS BUYER

i3 VERTICALS, LLC

NTD HOLDINGS, INC. AND GH HOLDCO, INC.

AS SELLERS

AND

DAVID GRAVES AND TORY HUMPHRIES

AS OWNERS

CLOSING DATE: April 3, 2019

EFFECTIVE TIME: 12:01 a.m. Nashville, Tennessee time on April 1, 2019

MEMBERSHIP INTEREST PURCHASE AGREEMENT

THIS MEMBERSHIP INTEREST PURCHASE AGREEMENT (this "<u>Agreement</u>") is dated as of April 3, 2019, and is effective as of April 1, 2019 at 12:01 a.m. Nashville, Tennessee time (the "<u>Effective Time</u>") by and among (i) i3-Bearcat, LLC, a Delaware limited liability company ("<u>Buyer</u>"), (ii) i3 Verticals, LLC, a Delaware limited liability company ("<u>Buyer</u>"), (ii) i3 Verticals, LLC, a Delaware limited liability company ("<u>ILLC</u>"), (iii) NTD Holdings, Inc., a Texas corporation ("<u>NTD Holdings</u>"), and GH Holdco, Inc., a Texas corporation ("<u>GH Holdco</u>", and together with NTD Holdings, "<u>Sellers</u>", and each a "<u>Seller</u>"), and (iv) David Graves, a resident of the State of Texas ("<u>Graves</u>"), and Tory Humphries, a resident of the State of Texas ("<u>Humphries</u>", and together with Graves, "<u>Owners</u>", and each an "<u>Owner</u>"). Collectively, Sellers, Owners and Buyer are referred to in this Agreement as the "<u>Parties</u>".

RECITALS:

A. NTD Holdings owns 100% of the membership interests of Northeast Texas Data, LLC, a Texas limited liability company, formerly The Northeast Texas Data Corporation, a Texas corporation ("<u>NET Data</u>"), and GH Holdco owns 100% of the membership interests of Graves Humphries Stahl, LLC, a Texas limited liability company, formerly Graves Humphries Stahl, LLC, a Texas corporation ("<u>GHS</u>" and, together with NET Data, the "<u>Companies</u>" and each a "<u>Company</u>").

B. Together, Owners own 100% of the issued and outstanding shares of capital stock of each Seller. Prior to the Restructuring (as defined below), Owners held all of the issued and outstanding shares of capital stock of the Companies.

C. The Companies are engaged in the business of developing, enabling, supporting, marketing, distributing and selling software products and services to small and midsize county governments and city court systems, including products and services focusing on solutions in the following categories: (1) county and district court management; (2) small court management; (3) financial management; (4) law enforcement; (5) property tax; (6) records management and (7) fee and fine collection services (collectively, the "Business").

D. Buyer desires to purchase from Sellers, and Sellers desire to sell to Buyer, all of the Equity Interests on the terms and subject to the conditions of this Agreement.

E. At least four (4) days prior to the Closing Date, Sellers and Owners effected the following restructuring (the "<u>Restructuring</u>"):

1

(i) Owners formed NTD Holdings and on the date of formation contributed all of the issued and outstanding shares of capital stock of NET Data to NTD Holdings in exchange for shares of capital stock of NTD Holdings;

(ii) Owners formed GH Holdco and on the date of formation contributed all of the issued and outstanding shares of capital stock of GHS to GH Holdco in exchange for shares of capital stock of GH Holdco;

(iii) Owners elected for NTD Holdings to be treated as an S corporation within the meaning of Section 1361(a) of the Code for U.S. federal and state income tax purposes effective on the day of its formation;

(iv) Owners elected for GH Holdco to be treated as an S corporation within the meaning of Section 1361(a) of the Code for U.S. federal and state income tax purposes effective on the day of its formation;

(v) NTD Holdings and GH Holdco each filed a Form 8869 Qualified Subchapter S Subsidiary Election electing (each a "<u>QSub Election</u>") for NET Data and GHS, respectively, to each be treated as a QSub effective on the day of the formation of NTD Holdings and GH Holdco, respectively; and

(vi) NET Data and GHS each converted (each a "<u>Disregarded Entity Conversion</u>") from a corporation to a limited liability company, effective at least two (2) days after the effective dates of the QSub Elections for NTD Holdings and GH Holdco, respectively.

F. The Parties desire to enter into this Agreement for the purpose of setting forth their mutual understandings and agreements with respect to the foregoing.

NOW, THEREFORE, intending to be legally bound and in consideration of the foregoing premises, the representations and warranties, mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties covenant and agree as follows:

ARTICLE 1 - DEFINITIONS

For purposes of this Agreement, the capitalized terms and variations thereof not otherwise defined in the body of this Agreement shall have the meanings ascribed to them in <u>Schedule 1</u> attached hereto.

ARTICLE II - PURCHASE AND SALE

2.1 <u>Purchase and Sale of Equity Interests</u>. As of the Effective Time, Sellers hereby sell, transfer and convey all of the Equity Interests to Buyer, free and clear of all Liens other than Permitted Encumbrances, in exchange for the Purchase Price.

2.2 <u>Purchase Price</u>. The aggregate consideration for the purchase and sale of the Equity Interests, subject to adjustment pursuant to <u>Section 2.3</u>, shall be as follows (collectively, the "<u>Purchase Price</u>"):

(a) Buyer shall pay to Sellers an aggregate amount equal to Twenty Four Million Dollars (\$24,000,000), adjusted in accordance with <u>Section 2.3</u> (the "<u>Closing Cash Purchase Price</u>"), by wire transfer of immediately available funds; <u>plus</u>

2

- (b) the Gross Up Payment, as calculated in accordance with <u>Section 6.2(j)</u>; <u>plus</u>
- (c) the Contingent Consideration.

2.3 Purchase Price Adjustment; Working Capital Matters.

(a) No less than one (1) Business Day prior to the Closing, Sellers shall prepare and deliver, or cause the Companies to prepare and deliver, to Buyer an officer's certificate of the Companies (the "<u>Estimated Closing Statement</u>") that contains a good faith estimate, prepared in accordance with the Historical Accounting Principles, of (i) the Net Working Capital (the "<u>Estimated Net Working Capital</u>"), (ii) the amount of Indebtedness of the Companies calculated through and including the Closing that will be unpaid immediately prior to the Closing (including final bills and wire transfer instructions as applicable) (the "<u>Estimated Indebtedness</u>"), and (iii) the Transaction Expenses calculated through and including the Closing that will be unpaid immediately prior to the Closing, plus the aggregate amount of the Transaction Expenses that will become payable after the Closing, to the extent calculable (the "<u>Estimated Transaction Expenses</u>"). If the Estimated Net Working Capital reflects Current Assets in excess of Current Liabilities, then the Closing Payment payable to Sellers at the Closing pursuant to <u>Section 2.6(a)</u> shall be increased by such excess amount. If the Estimated Net Working Capital reflects Current Liabilities in excess of Current Assets, then the Closing Payment payable to Sellers at the Closing pursuant to <u>Section 2.6(a)</u> shall be decreased by such excess amount.

(b) No later than ninety (90) calendar days after the Closing Date, Buyer shall prepare in good faith and deliver to Representative a statement (the "<u>Closing_Statement</u>") setting forth Buyer's calculation, in reasonable detail and prepared in accordance with GAAP, of (i) the Net Working Capital (the "<u>Net Working Capital Calculation</u>"), (ii) the amount of Indebtedness of the Companies calculated through and including the Closing and unpaid immediately prior to the Closing (including final bills and wire transfer instructions as applicable) (the "Closing Indebtedness"), (iii) the Transaction Expenses calculated through and including the Closing and unpaid immediately prior to the Closing, plus the aggregate amount of the Transaction Expenses paid or payable after the Closing, to the extent calculable (the "Closing Transaction Expenses"), and (iv) Buyer's proposed calculation of the Adjustment Calculation. If Buyer has not prepared and delivered to Representative the Closing Statement by the conclusion of the ninetieth (90th) day after the Closing Date (a "Buyer Adjustment Default"), then Buyer will have forfeited its right to prepare and deliver the Closing Statement, and Representative may, but is not required to, (x) prepare and deliver its own Closing Statement in good faith to Buyer not more than one hundred twenty (120) days after the Closing Date or (y) obtain an injunction, specific performance or other equitable relief to require Buyer to deliver its Closing Statement prepared in good faith, in which case all costs of such injunction, specific performance or other equitable relief will be borne solely by Buyer. If Representative elects to prepare the Closing Statement after a Buyer Adjustment Default, then such Closing Statement will be final and binding on the Parties for all purposes. Notwithstanding anything in this Agreement to the contrary, the remedies described in this Section 2.3(b) are Sellers' sole and exclusive remedies with respect to

3

any and all claims arising from Buyer's failure to prepare and deliver to Representative the Closing Statement by the conclusion of the ninetieth (90th) day after the Closing Date.

(c) On or prior to the thirtieth (30th) calendar day following Buyer's delivery of the Closing Statement to Representative, Representative may give Buyer a written notice stating in reasonable detail any and all of Sellers' objections (an "<u>Objection Notice</u>") to the Closing Statement or the determination of the Net Working Capital Calculation, the Closing Indebtedness, the Closing Transaction Expenses or the Adjustment Calculation as determined by Buyer. During such thirty (30) calendar-day period, Buyer will provide Representative and his accountants full access, during normal business hours and upon reasonable notice, to (i) review the financial books and records of the Companies, and (ii) the employees and other representatives of Buyer and the Companies who were responsible for the preparation of the Closing Statement or any items therein to respond to questions relating to the preparation of the Closing Statement. Any Objection Notice shall specify in reasonable detail the nature and dollar amount of the objections and the reasonable basis or bases therefor. Any determination set forth on the Closing Statement to which Representative does not object in the Objection Notice shall be deemed acceptable and shall be final and binding upon the Parties upon delivery of the Objection Notice. The failure by Representative to deliver an Objection Notice within such thirty (30) day period shall constitute Sellers acceptance of the Adjustment Calculation, which shall be final and binding on Sellers and Representative for all purposes of this Agreement.

(d) Following Buyer's receipt of any Objection Notice, Representative and Buyer shall attempt to negotiate in good faith to resolve such dispute. In the event that Representative and Buyer fail to agree on any of Representative's proposed adjustments set forth in the Objection Notice, within thirty (30) days after Buyer receives the Objection Notice, then Weaver (the "Independent Accountant") shall make the final, binding determination, absent fraud or manifest error, regarding the proposed adjustments set forth in the Objection Notice that are not resolved by Representative and Buyer (the "Adjustment Calculation Disputed Items"). Buyer, on the one hand, and Representative, on the other hand, each shall provide the Independent Accountant with their respective determinations of the Adjustment Calculation Disputed Items. The Independent Accountant shall make its determination of the Adjustment Calculation Disputed Items and the resultant Final Net Working Capital Calculation (defined below), Final Indebtedness (defined below), Final Transaction Expenses (defined below) and Final Adjustment Calculation which determination shall be final and binding on Sellers and Buyer. The determination of any of the Adjustment Calculation Disputed Items by the Independent Accountant shall be within, and limited by, the range comprised of the respective determination of each of Buyer's and Representative's calculation with respect to such Adjustment Calculation Disputed Items. The determination of the Adjustment Calculation Disputed Items by the Independent Accountant shall be based on whether such Adjustment Calculation Disputed Items have been calculated in accordance with the standards set forth in this Section 2.3, and the Independent Accountant is not to make any other determination. The Independent Accountant shall make its determination based solely on presentations and supporting material provided by Buyer and Representative and not pursuant to any independent

4

review. The fees, costs and expenses of the Independent Accountant shall be split equally between Sellers and Buyer. Any of the Parties may require that the Independent Accountant enter into a customary form of confidentiality agreement with respect to the work papers and other documents and information provided to the Independent Accountant under this <u>Section 2.3</u>. As used herein, "<u>Final Net Working Capital Calculation</u>" means the Net Working Capital Calculation as ultimately determined in accordance with <u>Section 2.3(c)</u> or this <u>Section 2.3(d)</u>, as applicable, "<u>Final Indebtedness</u>" means the amount of Indebtedness as ultimately determined in accordance with amount of Transaction Expenses as ultimately determined in accordance with <u>Section 2.3(c)</u> or this <u>Section 2.3(c)</u> or this <u>Section 2.3(c)</u> or this <u>Section 2.3(c)</u> or this <u>Section 2.3(d)</u>, as applicable, "<u>Final Indebtedness</u>" means the amount of Indebtedness as ultimately determined in accordance with <u>Section 2.3(d)</u>, as applicable.

(e) If the Final Adjustment Calculation is a negative number after final determination pursuant to this <u>Section 2.3</u>, then Owners shall, on behalf of Sellers (in accordance with each Owner's Percentage Interest), pay to Buyer an amount equal to the Final Adjustment Calculation by wire transfer of immediately available funds to an account designated in writing by Buyer. If the Final Adjustment Calculation is a positive number after final determination pursuant to this <u>Section 2.3</u>, then Buyer shall pay to Owners, on behalf of Sellers (in accordance with each Owner's Percentage Interest), by wire transfer of immediately available funds an amount equal to the Final Adjustment Calculation. Any amount due under this <u>Section 2.3(e)</u> shall be paid within five (5) Business Days after the date of final determination pursuant to this <u>Section 2.3</u>. The Purchase Price shall be deemed to be increased or decreased, as applicable, by the Final Adjustment Calculation, as determined in accordance with this <u>Section 2.3</u>.

2.4 <u>The Closing</u>. The Closing shall be deemed to occur contemporaneously with the Parties' execution and delivery of this Agreement. The date of the Closing shall be the "<u>Closing Date</u>" for purposes of this Agreement. The Closing shall take place remotely via the electronic exchange of documents and signatures on the Closing Date.

2.5 <u>Actions of Sellers at Closing</u>. At or prior to the Closing, unless otherwise waived in writing by Buyer, Sellers shall deliver to Buyer the following:

(a) a certificate of existence of each Company from the Secretary of State of Texas and a certificate of tax clearance for each Company from the Texas Comptroller of Public Accounts, each dated within ten calendar days of the Closing Date;

(b) fully-executed resignations of the directors and officers of the Company;

(c) Employee Non-Competition, Non-Solicitation, Confidentiality and Inventions Agreements, executed by Humphries and Dave Graves (i.e., the son of Graves);

(d) letters containing an offer of employment by Buyer or one of its affiliates to each of Humphries and Dave Graves (the "<u>Offer Letters</u>"), executed by Humphries and Dave Graves;

67350961.11

(e) a Closing Memorandum providing for payments of all amounts owed between the Parties or to third parties at the Closing, including the Closing Payment (the "<u>Closing Memorandum</u>"), executed by each Seller;

(f) evidence of the satisfaction of all payment obligations for Transaction Expenses and Indebtedness of the Companies outstanding as of the Closing Date (including any interest, prepayment premiums or penalties and other fees and charges) or evidence of the arrangement of Sellers or the Companies to satisfy such payment obligations on the Closing Date pursuant to the terms of this Agreement, including true, correct and complete invoices or releases reflecting all Transaction Expenses and payoff letters or similar releases with respect to such Indebtedness, and the release of any Liens on the properties and assets of the Companies (other than Permitted Encumbrances) and the termination of all UCC financing statements which have been filed with respect to such Indebtedness;

(g) the Estimated Closing Statement;

(h) actual or constructive possession of the Books and Records of the Companies;

(i) assignment agreements evidencing the transfer of the Equity Interests to Buyer, duly executed by each Seller;

(j) a real property lease for the Business's location at 1110 Enterprise Drive, Sulphur Springs, Texas 75482 (the "Lease"), executed by Net Data Partners, as landlord; and

(k) such other documents as may be reasonably required by Buyer to effectuate the Contemplated Transactions.

Simultaneously with the delivery of the foregoing items, Seller will take all such steps as may reasonably be required to put Buyer in actual possession and operating control of the Equity Interests.

2.6 <u>Actions of Buyer at Closing</u>. At or prior to the Closing, unless otherwise waived in writing by Sellers, Buyer shall deliver to Sellers the following:

(a) an amount in cash, by wire transfer of immediately available funds, equal to (i) the Closing Cash Purchase Price, <u>minus</u> (ii) the sum of the Transaction Bonuses (including the Companies' portion of any employment Taxes related thereto), if any, <u>minus</u> (iii) the sum of the Estimated Transaction Expenses, <u>minus</u> (iv) the sum of the Estimated Indebtedness, and adjusted to account for the Estimated Net Working Capital in accordance with <u>Section 2.3</u>, to the accounts and in the allocation set forth in the Closing Memorandum (the "<u>Closing Payment</u>");

(b) a Certificate of the Secretary of Buyer, certifying (i) the resolutions of the sole member of Buyer, authorizing and approving the performance of the Contemplated Transactions and the execution and delivery of this Agreement and the documents described herein, and (ii) incumbency for the officers of Buyer executing this Agreement, making

6

certifications for Closing or executing agreements or instruments contemplated hereby dated as of the Closing Date;

(c) the Offer Letters, executed by Buyer;

(d) the Closing Memorandum, executed by Buyer;

(e) the Lease, executed by Buyer;

(f) 200,000 Options, evidenced by i3V's standard stock option award agreement, to be allocated between certain key employees of the Companies identified by Sellers in writing to Buyer prior to Closing, such Options to be delivered directly by Buyer or i3V to such employees in connection with the post-Closing employment onboarding process; and

(g) Such other documents as may be reasonably required by Sellers to effectuate the Contemplated Transactions.

2.7 Contingent Consideration.

(a) Sellers may receive additional consideration from Buyer based on the post-Closing performance of the Acquired Business (the "<u>Contingent Consideration</u>") in accordance with this <u>Section 2.7</u> and <u>Section 2.8</u>. Within ten (10) calendar days following the final determination of achievement of the Performance Threshold (as defined in <u>Exhibit 2.7</u>), Buyer shall pay the Contingent Consideration by wire transfer of immediately available funds to Sellers in the same manner that the Closing Payment was paid. The Parties agree to report any Contingent Consideration paid to Sellers for Tax and accounting purposes as Purchase Price consistent with this Agreement.

(b) If, as of the last day of the Measurement Period, the Performance Metric generated by the Acquired Business during the Measurement Period exceeded the Performance Threshold (as defined in <u>Exhibit 2.7</u>), then Sellers will be entitled to be paid aggregate Contingent Consideration of \$1,000,000 by Buyer.

(c) During the Measurement Period, without the prior written consent of Sellers, Buyer shall not take any action or omit to take any action with the willful intent or purpose of reducing or delaying payment of the Contingent Consideration.

(d) During the Measurement Period, Buyer shall maintain a separate profit and loss statement for Companies' revenues and expenses.

(e) The Parties acknowledge and agree that the express provisions of this <u>Section 2.7</u> and <u>Section 2.8</u> will solely govern their contractual relationship with respect to the Contingent Consideration.

(f) Notwithstanding anything in this <u>Section 2.7</u> to the contrary, no Contingent Consideration shall be due for so long as any Seller or Owner is or remains in material breach of this Agreement or any of the other Transaction Documents. This <u>Section</u> 2.7(f)

7

is not subject to the provisions of <u>Section 2.8</u> and any disputes arising under this <u>Section 2.7(f)</u> shall be subject to the dispute resolution provisions described in <u>Section 11.6</u>.

2.8 Contingent Consideration Dispute Resolution.

(a) No later than 45 days after the end of the Measurement Period, Buyer shall calculate the Performance Metric for the Measurement Period, and shall deliver a written notice to Sellers setting forth the calculation of the Performance Metric and the Contingent Consideration owed to Sellers. Within 30 days after receipt of Buyer's calculations, Sellers may, during regular business hours and at their own cost and expense, inspect and audit such books, records and other work papers of Buyer as are reasonably necessary to verify that Buyer has calculated the Performance Metric in compliance with the terms and conditions of this Agreement.

(b) If Sellers dispute Buyer's determination with respect to the achievement of the Performance Threshold, Sellers shall notify Buyer in writing by delivery of a notice within 30 days of receiving Buyer's calculation (a "<u>Contingent Consideration Dispute Notice</u>"), which Contingent Consideration Dispute Notice shall set forth in reasonable detail the basis for such dispute. In the event of such a dispute, Sellers and Buyer shall work in good faith to resolve the dispute. If, after ten (10) Business Days, a dispute still exists, such dispute shall be submitted the Independent Accountant. The Independent Accountant shall make its determination regarding the disputed items as promptly as practicable, and such determination shall be the final determination. The Independent Accountant shall determine whether the Contingent Consideration has been earned. Any expenses relating to the engagement of the Independent Accountant shall be borne by the Party whose determination of whether Contingent Consideration has been earned was incorrect. The Independent Accountant shall be instructed to use every reasonable effort to perform its services within fifteen (15) Business Days of the Independent Accountant's engagement and, in any case, as soon as practicable after submission thereof. The Independent Accountant's decision shall be final and binding on the Parties. The Parties shall make available to the Independent Accountant, as applicable, such books, records and other information as the Independent Accountant may reasonably request. In the event such the Independent Accountant providers an award hereunder, the Party owing such award shall pay all such award, by wire transfer of immediately available funds.

(c) In the event that actions are taken or decisions are made by Buyer or an affiliate of Buyer that adversely affect the achievement of the Performance Threshold, and such actions or such decisions are determined by the Independent Accountant to have been made by Buyer or an affiliate of Buyer in bad faith and for the primary purpose of frustrating the achievement of the milestone, then the milestone will be deemed to have been achieved and the Contingent Consideration will be paid to Sellers.

(d) Additionally, in the event Buyer or any Company (i) sells or otherwise disposes of the Acquired Business, (ii) sells or otherwise disposes of all or substantially all of its assets (whether by asset purchase, equity purchase, merger, combination, joint venture or otherwise) or (iii) substantially ceases or discontinues operations of the Acquired Business, in each case, before the final determination of the achievement of the Performance Threshold in

8

accordance with this <u>Section 2.8</u>, then Buyer shall pay the Contingent Consideration to Sellers as if it had been fully earned during the Measurement Period. For the avoidance of doubt, this <u>Section 2.8(d)</u> shall not apply to any transfer of the Acquired Business by Buyer to any controlled affiliate of Buyer.

ARTICLE III - REPRESENTATIONS AND WARRANTIES OF OWNERS CONCERNING THEMSELVES AND SELLERS

Each Seller and Owner represents and warrants to Buyer as of the Effective Time and the Closing Date as follows:

3.1 Authorization; Binding Effect; Status.

(a) Such Owner and each Seller has full power and authority to execute and deliver this Agreement and all other agreements, documents and instruments to be executed and delivered hereunder (the "<u>Transaction Documents</u>") and to perform its obligations hereunder and thereunder. The Transaction Documents and performance and consummation of the Contemplated Transactions are and have been approved and authorized by all requisite action of such Owner and Sellers, and no other Proceedings on the part of such Owner or Sellers are necessary therefor. The Transaction Documents have been duly executed and delivered by such Owner and Sellers and, assuming due authorization, execution and delivery of this Agreement and the other Transaction Documents by the other Parties, the Transaction Documents are the valid and legally binding obligation of such Owner and Sellers, enforceable against such Owner and Sellers in accordance with its terms, subject to applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting creditor's rights generally from time to time in effect and limitations on enforcement of equitable remedies.

(b) Each Seller (i) is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of Texas, and (ii) has all requisite corporate power and authority to engage in the Contemplated Transactions, carry on its business as now and heretofore conducted and to own, operate and lease its properties and assets and to perform its obligations under this Agreement.

3.2 <u>Non-contravention</u>. Neither the execution and delivery of this Agreement nor any of the Transaction Documents to which such Owner or any Seller is a party, nor the consummation of the Contemplated Transactions, will: (a) violate any Law, injunction, judgment, ruling, charge, or other restriction of any Governmental Authority to which such Owner or any Seller is subject; (b) conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify, or cancel, or require any notice under any agreement, Contract, lease, license, instrument, or other arrangement to which such Owner is a party or by which it is bound or to which any of its assets or properties are subject; or (c) result in the imposition or creation of any Liens upon or with respect to the Equity Interests or any assets or properties of the Company. No spouse or former spouse of such Owner has any claim, right, or power under or as a result of any community or other marital property laws that will, or would reasonably be expected to, give such Person the legal right to prevent, delay, or render invalid such Owner's execution of this Agreement or any of the

9

Transaction Documents to which such Owner is a party or the consummation of the Contemplated Transactions or to cause any Liens to exist upon the Equity Interests or any assets or properties of the Company at the Effective Time or at any time thereafter. No other Person has any claim, right, or power under or as a result of any contractual right or otherwise that will, or would reasonably be expected to, give any such Person the legal right to prevent, delay, or render invalid such Owner's or any Seller's execution of this Agreement or any of the Transaction Documents to which such Owner or any Seller is a party or the consummation of the Contemplated Transactions or to cause any Liens to exist upon the Equity Interests or any assets or properties of the Company at the Effective Time or at any time thereafter.

3.3 <u>Necessary Consents</u>. Neither such Owner nor any Seller is a party to or bound by any Lien or Contract or instrument, or any Law that requires a Required Consent.

3.4 <u>Title to Equity Interests</u>. As of the date hereof, each of NTD Holdings and GH Holdco has sole voting power and sole power of disposition, in each case with respect to all of the Equity Interests held by such Seller with no limitations, qualifications or restrictions on such rights and powers (other than Permitted Encumbrances). The Equity Interests are free and clear of any Liens (other than Permitted Encumbrances). Neither Seller nor such Owner is subject to any agreements, arrangements, options, warrants, calls, rights, commitments or other restrictions relating to the sale, transfer, purchase, redemption or voting of its Equity Interests. None of the Companies, any Seller, or such Owner has granted to any Person any right of first refusal, preemptive right, subscription right or similar right with respect to the Equity Interests.

3.5 <u>Capitalization of Sellers</u>. As of the date hereof, each Owner is the sole record and sole beneficial holder of 100% of its respective portion of the issued and outstanding shares of capital stock of Sellers. Together, Owners own 100% of the issued and outstanding shares of capital stock of Sellers.

ARTICLE IV - REPRESENTATIONS AND WARRANTIES OF OWNERS CONCERNING THE COMPANIES

Owners represent and warrant to Buyer as of the Effective Time and the Closing Date, except as and subject to the disclosures set forth in the Disclosure Schedules (provided that, no matter disclosed in one section of the Disclosures Schedules will be deemed disclosed in another section of the Disclosure Schedules unless it is reasonably apparent on its face that the matter is responsive to such other representation), as follows:

4.1 <u>Status</u>. Each Company (i) is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Texas, (ii) has all requisite limited liability company power and authority to carry on its business as now and heretofore conducted and to own, operate and lease its properties and assets and to perform its obligations under the Contracts, and (iii) is duly qualified or licensed to transact business in and is in good standing under the Laws of each jurisdiction where such qualification is required, except where failure to obtain such licensing or qualification would not reasonably be expected to have a Material Adverse Effect. Owners have delivered to Buyer copies of the Articles of Organization and Company Agreement of each Company, as currently in effect.

67350961.11

4.2 Capitalization of Companies.

(a) <u>Schedule 4.2(a)</u> accurately sets forth the authorized, issued and outstanding membership interests, ownership interests, transferable interests or other securities of each Company and the name and number of such securities held by each owner thereof. All of the Equity Interests have been duly authorized, are validly issued, fully paid and nonassessable, and are owned of record and beneficially by Sellers, free and clear of all Liens (other than Permitted Encumbrances), and are not subject to any preemptive, subscription or similar rights that will survive the Closing. Other than the Equity Interests, there are no other ownership interests, transferable interests, securities, participations or other equivalents (however designated and whether voting or nonvoting) of the Companies issued or outstanding. No current or former equityholder of the Companies or any other Person is contesting the ownership of the Equity Interests or any dividends, distributions or contributions relating thereto. Except for this Agreement, there are no outstanding or authorized options, warrants, rights, Contracts, pledges, calls, puts, rights to subscribe, redeem, repurchase or otherwise acquire, conversion rights or other agreements, commitments or obligations (contingent or otherwise) to which any of the Companies are parties or which are binding upon any of the Companies providing for the issuance, sale, disposition or acquisition of any equity or any rights or interests exercisable therefor. There are no outstanding or authorized equity appreciation, phantom equity or similar rights with respect to any of the Companies. The Companies have not violated any foreign, federal or state securities or "blue sky" Laws in connection with the offer, sale or issuance of their securities (including the Equity Interests). There are no Contracts to which a Company or a Seller is a party with respect to the voting or transfer of the Equity Interests or any other securities of any of the Companies. There are no bonds, debentures, notes or other indebtedness of the Companies having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which any shareholder of a Company may vote. Scott Stahl holds no financial, governance or other rights in any of the Companies.

(b) None of the Companies owns, directly or indirectly, (i) any subsidiaries, (ii) with respect to any Person that is a corporation, any shares, interests, participations or other equivalents (however designated and whether voting or nonvoting) of capital stock of such Person or any right, warrant or option to acquire any of the foregoing, or (iii) with respect to any Person that is not a corporation, any general partnership interests, limited partnership interests, membership or limited liability interests, beneficial interests or other similar interests of or in such Person (including any common, preferred or other interest in the capital or profits of such Person, and whether or not having voting or similar rights), or any right, warrant or option to acquire any of the foregoing.

4.3 <u>Required Consents</u>. Except as described in <u>Schedule 4.3</u> and other than with respect to Permitted Encumbrances, no Company is a party to or bound by any Lien or Contract or instrument, or any Law that requires the consent of any other party to the execution of this Agreement or the consummation of the Contemplated Transactions, gives rise to a right of first refusal in favor of any other party as a result of the execution of this Agreement or the consummation of the Contemplated Transactions of the Contemplated Transactions or prohibits or requires the consent or

67350961.11

notification of another to, any of the Contemplated Transactions (any exceptions to the representations and warranties contained in this <u>Section 4.3</u> being the "<u>Required Consents</u>").

4.4 No Conflict. The negotiation, execution, delivery and performance of the Transaction Documents (or any letter of intent related thereto), consummation of the Contemplated Transactions and compliance with any of the provisions thereof will not (a) assuming that all Required Consents have been obtained, violate any Law to which any of the Companies or the Business may be subject, (b) conflict with or result in a breach of any provision of the organizational documents of any of the Companies, (c) other than as provided on <u>Schedule 4.3</u>, require any consent, approval or authorization of, or notice to, or declaration, filing or registration with, any Governmental Authority that is required by Law or the regulations of any Governmental Authority, (d) violate any Order to which any Company or the Business may be subject, (e) contravene, conflict with or result in a violation of any of the terms or requirements of, or give any Governmental Authority the right to revoke, withdraw, suspend, cancel, terminate or modify, any License that is held by and that is material to any Company or that otherwise relates and is material to the Business, (f) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice or lapse of time or both would constitute a default) under, or result in the termination or in a right of termination or cancellation of, or accelerate the performance required by, or result in the creation of any Lien upon any assets of any Company or result in being declared void, voidable, without further binding effect, or subject to amendment or modification any of the terms, conditions or provisions of, any Contract, license, franchise, permit, or other instrument or commitment or obligation to which any Company may be bound or affected, (g) result in the termination, denial or impairment of any Contract, license, franchise, permit, or other instrument or commitment or obligation to which any Company may be bound or affected, or (h) cause any Company's assets to be reassessed or revalued by any Governmental Authority or other Governmental Authority.

4.5 Financial Statements; Company Receivables.

(a) <u>Schedule 4.5(a)</u> includes copies of the following financial statements of each Company (collectively, the "<u>Financial Statements</u>"):

(i) Unaudited balance sheets, income statements and statements of cash flows for the past three (3) fiscal years of each Company as of the end of each such fiscal year and any notes thereto;

<u>Date</u>"); and,

(ii) Unaudited balance sheet (the "<u>Interim Balance Sheet</u>"), dated as of January 31, 2019 (the "<u>Balance Sheet</u>

(iii) Unaudited income statement and statement of cash flow for the period beginning January 1, 2019 and ending on the Balance Sheet Date.

(b) The Financial Statements fairly present in all material respects the financial condition, cash flows and results of operations of each Company at the respective dates thereof and for the periods referred to therein. The Financial Statements have been prepared in accordance with the Historical Accounting Principles applied on a consistent basis throughout

12

the periods indicated. The Financial Statements have been prepared from and are in accordance with the Books and Records of each applicable Company (including but not limited to invoices, settlement reports and other reports received from and provided by the Processors).

(c) Each Company has paid, as of the Closing, all amounts due pursuant to prior merger agreements or purchase agreements entered into by an Owner, a Company or any predecessor (a list of which, including the amount paid pursuant to such purchase agreements, future amounts that are or may be due, and any related conditions or contingencies, is set forth as <u>Schedule 4.5(c)</u>) and, to the Knowledge of any Owner, there is no dispute or claim, whether on behalf of a Company or the counterparty to such purchase agreements, related to consideration, attrition, indemnification or other matters related to such purchase agreements.

(d) Since December 31, 2018, each Company has paid its payables only in the ordinary course of business, consistent with past practice. Since December 31, 2018, no Company has made any purchase, made any payment, or funded any Liability outside of the ordinary course of business.

(e) No Owner or Seller has a right to receive (i) payments from any Customers, Referral Sources, processors, or any other third parties in connection with the Customer Agreements, Referral Source Agreements or any Contracts relating to any Company or the Business, or (ii) any payments owed by third parties with respect to any Company or the Business (whether or not related to the Customers or Referral Sources), including any convenience fees, commissions, ancillary fees, license fees, royalties and any other revenue streams relating to transactions of the Customers, including all amounts relating to collections recovery, retrieval fees, miscellaneous credits and other similar funds.

(f) All notes and accounts receivable reflected on the Interim Balance Sheet of each Company (i) are reflected properly on the Financial Statements, (ii) are bona fide receivables representing amounts due with respect to actual, arm's length transactions entered into in the ordinary course of business for goods sold and delivered or services performed, and (iii) are legal, valid and binding obligations of the obligors, in accordance with their terms and are not subject to any counterclaim or set-off other than normal cash discounts accrued in the ordinary course of business consistent with past practice. No request for deduction, discount or concession has been made in respect of any notes or accounts receivable reflected on the Interim Balance Sheet.

4.6 <u>Undisclosed Liabilities</u>. Except as set forth on <u>Schedule 4.6</u>, no Company has any Liabilities of any nature, accrued, absolute, fixed, contingent, liquidated, unliquidated, or otherwise and whether or not due, other than Liabilities that (a) were reflected or reserved against on the Financial Statements or (b) have been incurred in the ordinary course of business consistent with past practice since the Balance Sheet Date and which are not, individually or in the aggregate, material in amount.

4.7 <u>Absence of Certain Changes</u>. Since the Balance Sheet Date, (a) each Company has conducted its business only in the ordinary course of business, consistent with past practice and in compliance with Law in all material respects, (b) no Company has amended the terms of

67350961.11

or terminated any Material Contracts, and (c) no Company has experienced a Material Adverse Effect, and except as set forth on <u>Schedule 4.7</u>, to the Knowledge of any Owner, no circumstance exists that can reasonably be expected to result in a Material Adverse Effect.

4.8 <u>Brokers or Finders</u>. Except for Waterview Investment Banking, Inc., no Owner, Company or any of their respective agents has engaged any finder or broker or incurred any Liability, contingent or otherwise, for brokerage or finders' fees or agents' commissions in connection with this Agreement and the Contemplated Transactions.

4.9 Real Property and Personal Property.

(a) Except as set forth on <u>Schedule 4.9(a)(i)</u>, no Company currently owns, and for the past three years has not owned, any Real Property. All Real Property used by either Company is leased pursuant to those real estate leases set forth on <u>Schedule 4.9(a)(ii)</u> (the "<u>Real Property Leases</u>"). No Company is in breach of any material term or condition of any Real Property Lease, nor, to the Knowledge of any Owner, is any Company in violation of any applicable Law, condemnation, assessment or any similar action, relating to any Real Property or the operation thereof.

(b) All Personal Property is in good operating condition and repair (subject to normal wear and tear). As of the Closing, all Personal Property that is owned by each Company will be free and clear of any Liens other than Permitted Encumbrances. No Person other than a Company owns any Personal Property situated on the Real Property, except for (i) items leased or licensed by a Company or improvements to items leased or licensed by a Company, and (ii) personal property of Owners or a Company's employees or visitors which is not required for the operation of the Business.

4.10 Contracts.

(a) <u>Schedule 4.10</u> is a true and complete list of all of the following material Contracts of each Company (the "<u>Material Contracts</u>"):

Company;

(i) all written or oral employment or consulting Contracts pursuant to which services are rendered to a

(ii) all Contracts under which a Company is or the Business or Buyer will after the Closing be restricted from carrying on any business or other activities anywhere in the world;

(iii) all Contracts to purchase, lease, or sell assets or services involving aggregate annual payments in excess

of \$25,000;

(iv) all Contracts (including organization, partnership and joint venture agreements) under which (A) a Company has any Liability or obligation for debt or constituting or giving rise to a guarantee of any Liability or obligation of any Person or (B) any Person has any Liability or obligation constituting or giving rise to a guarantee of any Liability or obligation

14

of a Company, or any Liability or obligation to a Company, in each case involving any debt or Liability in excess of \$25,000 individually or \$50,000 in the aggregate;

(v) the Real Property Leases;

(vi) all Contracts with Processors, sponsor banks, independent sales organizations, Referral Sources or Material Customers;

¢25 000.

(vii) all Contracts involving aggregate annual consideration payable to or from a Company in excess of

\$25,000;

(viii) all Intellectual Property Contracts;

(ix) all Contracts which grant a third party a right of exclusive dealing with a Company, a right of first refusal, right of first offer, or similar option right, for any of the Equity Interests or the assets of any Company; and

(x) all Contracts necessary to operate, or are otherwise material to, the Business as it is currently being

conducted.

(b) No material breach or default in performance by a Company under any of the Material Contracts has occurred or is continuing, and, to the Knowledge of any Owner, no event has occurred, which with notice or lapse of time or both would constitute such a material breach or default. No Company has given or received from any other Person any notice or other communication regarding any actual, alleged or potential material breach or default under the Material Contracts. To the Knowledge of any Owner, no material breach or default by any other Person under any of the Material Contracts has occurred or is continuing, and no event has occurred which with notice or lapse of time or both would constitute such a material breach or default.

(c) There are no pending renegotiations of, or attempts to renegotiate, any material amounts paid or payable to or by a Company under any Contracts.

4.11 Intellectual Property; Computer Software.

(a) Company Intellectual Property.

(i) <u>Status of Owned Intellectual Property and Licensed Intellectual Property</u>. The Company Intellectual Property consists exclusively of the Owned Company Intellectual Property and Licensed Company Intellectual Property and constitutes all material Intellectual Property necessary and sufficient for each Company to conduct its Business (including the design, development, manufacture, use, sale, licensing or other exploitation of Company Products) as currently conducted. None of the Contemplated Transactions will alter, impair or otherwise adversely affect any rights of any Company in any Company Intellectual Property.</u>

15

(ii) <u>Owned Company Intellectual Property</u>. <u>Schedule 4.11(a)(ii)</u> sets forth a true, correct and complete list of all Owned Company Intellectual Property, including (A) Trademarks used including a list of applicable jurisdictions and the nature of the goods and services offered under the Trademarks and the dates of first use, (B) Domain Names used by each Company and, for each Domain Name, the registration date, any expiry date and name of registry, (C) all threatened or actual legal Proceedings (including reexamination and reissue Proceedings) before any court, tribunal or other Governmental Authority (including the World Intellectual Property Organization or equivalent authority anywhere in the world) related to any Company Intellectual Property Registrations and (D) any actions that must be taken within ninety (90) days after the date hereof for the purposes of obtaining, maintaining, perfecting, preserving or renewing any Company Intellectual Property Registrations, including the payment of any registrations is valid, enforceable and subsisting, all necessary registration, maintenance and renewal fees currently due in connection with such Company Intellectual Property Registrations have been made and all necessary documents, recordations and certificates in connection with such Company Intellectual Property Registrations have been filed with the relevant Governmental Authorities in the United States or other jurisdictions, as the case may be, for the purposes of prosecuting, perfecting and maintaining such Company Intellectual Property Registrations.

(iii) <u>Title to Owned Company Intellectual Property</u>. The Companies, in the aggregate, own, and have good and exclusive title to, all Owned Company Intellectual Property free and clear of any Lien and all Owned Company Intellectual Property is fully paid and not subject to any payments. All Owned Company Intellectual Property is fully transferable, alienable and licensable by the applicable Company and can be amended and modified, in each case, without restriction and without payment of any kind to any Person. Each Company has the right to use, practice and exploit all of its Owned Company Intellectual Property in the manner in which the foregoing Intellectual Property is being used, practiced or exploited or has, within the last five (5) years, been used, practiced or exploited.

(iv) <u>Ownership of Owned Company Intellectual Property</u>. No Person other than a Company has any ownership interest in or rights to any Company Products or any Owned Company Intellectual Property or any improvements made by or for any Company.

(v) <u>Licensed Company Intellectual Property</u>. <u>Schedule 4.11(a)(v)</u> sets forth a true, correct and complete list of all Licensed Company Intellectual Property. Each Company has a valid and enforceable license to use, practice and exploit all of its Licensed Company Intellectual Property in the manner in which the Licensed Company Intellectual Property is being used, practiced or exploited or has, within the last five (5) years, been used, practiced or exploited.

(b) Company Intellectual Property Contracts.

(i) <u>Inbound Intellectual Property Contracts</u>. <u>Schedule 4.11(b)(i)</u> sets forth a true, correct and complete list of all Persons who have licensed any Intellectual Property to any Company or granted to any Company any covenant not to sue or right with respect to any

67350961.11

Intellectual Property, including a description of the Contract, written or oral, and the Intellectual Property licensed by such Person ("<u>Inbound Intellectual Property Contracts</u>"). Sellers and the Companies have made available to Buyer true, correct and complete copies of all written Inbound Intellectual Property Contracts.

(ii) <u>Outbound Intellectual Property Contracts</u>. <u>Schedule 4.11(b)(ii)</u> sets forth a true, correct and complete listing of all Contracts, written or oral, with respect to any Company Intellectual Property licensed by any Company to any Person or pursuant to which any Company grants to any Person any covenant not to sue or right with respect to any Company Intellectual Property, in each case that is currently in effect ("<u>Outbound Intellectual Property Contracts</u>"). None of the Outbound Intellectual Property Contracts (A) cover or apply to any Intellectual Property of any affiliates of any Company and (B) would, after the Closing, apply to any Company, Buyer or any of their respective affiliates. Sellers and the Companies have made available to Buyer true, correct and complete copies of all written Outbound Intellectual Property Contracts.

(iii) <u>No Consents</u>. Immediately following the Closing, the Companies will be permitted to exercise all of their rights (A) to Company Intellectual Property and (B) under all Intellectual Property Contracts, to the same extent each Company would have been able to had the Contemplated Transactions not occurred and without being required to pay any additional amounts or consideration other than fees, royalties or payments that a Company would otherwise have been required to pay had such transactions not occurred.

(c) Third Parties and Company Intellectual Property.

(i) <u>Compensation to Employees</u>. Each Company has paid, in full, all mandatory compensation to employees, contractors and consultants in relation to all Company Intellectual Property, and neither this Agreement nor any of the Contemplated Transactions will result in any further amounts being payable to any current or former employees, contractors or consultants of any Company in relation to any Company Intellectual Property.

(ii) <u>Developers</u>. <u>Schedule 4.11(c)(ii)</u> lists all Persons who have contributed materially to the conception, creation or development of any Company Product or Owned Company Intellectual Property, a description of the Person's contribution, the Company for which the contribution was provided, and a notation of whether or not such person is or was an employee or, alternatively, a contractor or consultant of such Company. No current or former employee, contractor or consultant of any Company has ever excluded any Intellectual Property from any assignment to such Company of all right, title and interest in and to all Intellectual Property conceived, created or developed by such Person during such Person's employment by or engagement with such Company (including the rights to transfer, license, amend and modify such Intellectual Property) executed by such Person in connection with such Person's employment by or engagement with a Company.

(iii) <u>Infringement</u>. No Company has received any notice or request from any Person for indemnification, compensation or reimbursement with respect to any claim of infringement, misappropriation, misuse or violation of any Intellectual Property, which notice

67350961.11

or request has not been finally resolved. To the Knowledge of any Owner, the conduct of the Business as currently and formerly conducted, and the Company Intellectual Property as currently or formerly owned, licensed or used by each Company has not infringed, misappropriated, diluted or otherwise violated, and have not, do not and will not infringe, dilute, misappropriate or otherwise violate, the Intellectual Property or other rights of any Person. To the Knowledge of any Owner, no Person has infringed, misappropriated, diluted or otherwise violated, or is currently infringing, misappropriating, diluting or otherwise violating, any Owned Company Intellectual Property or Licensed Company Intellectual Property which is exclusively licensed to a Company.

(iv) <u>Third Party Access</u>. To the Knowledge of any Owner, no Company Products or Company Intellectual Property has been supplied or made available (and no Company has agreed to supply or make available any Company Products) to any Person other than pursuant to the Outbound Intellectual Property Contracts entered into in the ordinary course of business with Customers or End Users.

(v) <u>Institutions</u>. No government, university, college, other educational institution, research center or nonprofit institution has ever been involved in the research and development of Company Intellectual Property or Company Products (collectively, "<u>Institutions</u>") or has provided or provides facilities or funding for the creation or development of any Owned Company Intellectual Property or Company Products. No Institutions have any rights in or with respect to any Owned Company Intellectual Property or Company Products or any developments of any Intellectual Property made by any current or former employee, contractor or consultant of any Company that relate in any manner to Owned Company Intellectual Property or any Company Product.

(vi) <u>Transfer or Lapse of Intellectual Property Rights</u>. No Company has (A) transferred ownership of, or granted any exclusive license or exclusive right under or with respect to, or authorized the retention of any exclusive right with respect to or joint ownership of, any Intellectual Property that is or to the Knowledge of any Owner, was at any time owned or purported to be owned by a Company, to or with any other Person or (B) to the Knowledge of any Owner, permitted a Company's rights in any Owned Company Intellectual Property to lapse or enter the public domain. After giving effect to the Contemplated Transactions, no current or former partner, director, stockholder, officer, employee, contractor or consultant of any Company will own or retain any rights to use any of the Owned Company Intellectual Property.

(d) <u>Company Products</u>. <u>Schedule 4.11(d)(i)</u> sets forth a true, correct and complete list of all Company Products, including title, whether it is used by Customers, most current version and release number, the date the most current version was put into service, and the date of the last major update. Each Company Product conforms in all material respects to the specifications and documentation therefor. <u>Schedule 4.11(d)(ii)</u> sets forth a true, correct and complete list of all (i) software applications owned by or developed by or for any Company ("<u>Company Software</u>") and (ii) software applications or other Intellectual Property not owned by any Company (including public domain materials) that are incorporated or embedded in or

67350961.11

bundled with or necessary for the use of any Company Software or any Company Products, together with a description of such software, and a designation as Company Software or software licensed from another Person.

(e) Software, Systems and Support.

(i) <u>Malware</u>. To the Knowledge of any Owner, none of Company Software or Company Products contains any virus, Trojan horse, worm or other software routines or hardware components designed to permit unauthorized access to or designed to disable, erase or otherwise harm software, hardware or data. Owner and Companies have made available to Buyer a true, correct and complete list of all reported errors or bugs that have had or could reasonably be expected to have a Material Adverse Effect on Company Software or Company Products.

(ii) <u>Company Data</u>. All data contained in any databases of, or maintained on behalf of, any Company and all other information and data compilations used by, or necessary for the business of, any Company is owned by such Company or is used by or on behalf of such Company pursuant to a valid Contract or applicable Law (including, applicable Laws with respect to privacy of personal information and all data collection and data mining practices employed by any Company).

(iii) <u>Company IT Systems</u>. All information technology and computer systems (including computer software, data files, source and object codes, tools, user interfaces, manuals and other specifications and documentation (and all know-how relating to the foregoing), information technology and telecommunication hardware and other equipment), relating to the transmission, storage, maintenance, organization, presentation, generation, processing or analysis of data and information, whether or not in electronic format, used in or necessary to the conduct of the Business of any Company (collectively, "<u>Company IT Systems</u>"), have been properly maintained to ensure proper operation, monitoring and use. Company IT Systems are in good working condition to effectively perform all information technology operations necessary to conduct the Businesses of each Company as they are currently being conducted. No Company has experienced within the past three (3) years any material disruption to, or material interruption in, the conduct of business attributable to a defect, bug, breakdown or other failure or deficiency of Company IT Systems. Each Company has taken reasonable measures to provide for the back-up and recovery of the data and information necessary to the conduct of the Businesses of each Company without material disruption to, or material interruption in, the conduct of the Businesses of each Company without material disruption to, or material interruption to, or material interruption to, or material interruption in the conduct of the Businesses of each Company without material disruption to, or material disruption to, or material interruption in, the conduct of the Businesses of each Company without material disruption to, or material interruption in, the conduct of the Businesses of each Company without material disruption to, or material interruption in, the conduct of the Businesses of each Company without material disruption to, or material interruption in, the conduct of the Businesses of each Company without mate

(iv) <u>Open Source Materials</u>. <u>Schedule 4.11(e)(iv)</u> sets forth a true, correct and complete list of all Open Source Materials currently used by any Company, including in development or testing of any Company Software or Company Products, and (A) identifies the Open Source License applicable thereto, (B) identifies, where available, a URL at which such Open Source Materials are available and at which such Open Source License is identified, (C) describes the manner in which such Open Source Materials were or are used, (D) states whether (and, if so, how) such Open Source Materials were modified by or for a Company, (E) states whether such Open Source Materials were distributed by or for a Company, (F) states whether

67350961.11

such Open Source Materials were used, offered or made available on a hosted or similar basis by or for a Company and (G) describes how such Open Source Materials are integrated with or interact with Company Software or Company Products or any portion thereof. Except as set forth on <u>Schedule 4.11(e)(iv)</u>, no Company has (x) incorporated or embedded any Open Source Materials into, or combined or linked any Open Source Materials with, any Company Software or Company Product, or (y) distributed any Open Source Materials in conjunction with or for use with any Company Software or Company Products. To the Knowledge of Company, no Company has used any Copyleft Materials in a manner that requires Company Software or Company Products, or any portion thereof, or any Company Intellectual Property, to be subject to Copyleft Licenses.

(v) <u>Source Code</u>. <u>Schedule 4.11(e)(v)(A</u>) sets forth a true, correct and complete list of all software, firmware or other technology components not owned by a Company that are incorporated in, provided with or otherwise used, held for use or contemplated to be used by a Company in connection with the development, use, exploitation, support or maintenance of any Company Software or Company Products. Except as set forth in <u>Schedule 4.11(e)(v)(B)</u>, no Company has disclosed or delivered to any escrow agent or any other Person, other than employees performing services on a Company's behalf, any of the source code for any Company Software or relating to any Company Intellectual Property. No Person has any right, contingent or otherwise, under any Contract or otherwise, to obtain access to or use any such source code. No event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time) will, or reasonably could be expected to, result in the delivery, license or disclosure of any source code to any Person after the Closing Date.

(f) Platform.

(i) <u>Platform Equipment</u>. <u>Schedule 4.11(f)</u> identifies as of the Closing all computer equipment, hardware and other Personal Property of that is necessary for the conduct of, or are primarily used for the benefit of, the Platform. The Companies, together, have all right, title and interest in and to the Platform.

(ii) <u>Restrictions</u>. Neither the Companies nor Owners are under any obligation of confidentiality or restriction of use which would be violated by the terms of the Contemplated Transactions or interfere with the Companies' post-Closing quiet enjoyment, free use and benefit of the Platform.

(iii) <u>Adverse Events</u>. No Owner has any Knowledge of any action or omission on the part of any Owner, any Company or any third party which could adversely affect the value or the validity of the Platform.

(iv) <u>Ownership</u>. No Owner will directly or indirectly, challenge any Company's ownership of or the validity of the Platform or any applicable registration or application for the registration thereof.

(g) <u>Result of the Contemplated Transactions</u>. Neither this Agreement nor any of the Contemplated Transactions will result in (i) any Person being granted rights or access to,

or the placement in or release from escrow of, any software source code or other Company Intellectual Property, (ii) Buyer, any Company or any of their respective affiliates granting to any Person any ownership interest in, or any license, covenant not to sue or right under or with respect to, any Intellectual Property or (iii) Buyer, any Company or any of their respective affiliates or any of their Intellectual Property rights being bound by, or subject to, any non-compete or other restriction on the operation or scope of their respective businesses or their use and exploitation of Company Products.

4.12 Title to Assets.

(a) The Companies have good and valid title to, or a valid leasehold interest in, all properties and assets owned or used by the Companies, free and clear of all Liens (other than properties and assets disposed of in the ordinary course of business since the Balance Sheet Date, and except for Permitted Encumbrances). <u>Schedule 4.12(a)</u> sets forth a true and correct list of all aforementioned properties and assets that have been disposed of by any Company since the Balance Sheet Date.

(b) All buildings, plants, structures, fixtures, machinery, equipment, vehicles and other tangible personal property or assets of the Companies are in good operating condition and repair, ordinary wear and tear excepted, are adequate for the uses to which they are being put and are not in need of maintenance or repairs except for ordinary, routine maintenance and repairs that are not material in nature or cost.

4.13 Compliance; Approvals.

(a) Each Company has all Licenses that are required by Law to carry on the Business currently being conducted and own and use its assets as currently used, each of which is set forth on <u>Schedule 4.13(a)</u>. Each Company is and has been during the past three (3) years in compliance in all material respects with each such License. To the Knowledge of any Owner, no event has occurred or circumstance exists that (with or without notice or the lapse of time) constitutes or would reasonably be expected to directly or indirectly in a violation of or failure to comply with any term or requirement of any License.

(b) Each Company is, and at all times during the past three (3) years has been, in compliance in all material respects with all Laws and during the past five (5) years has been in compliance in all respects with the Telephone Consumer Protection Act and similar laws imposed at the state level. To the Knowledge of any Owner, no event has occurred or circumstance exists that (with or without notice or lapse of time) (i) constitutes or would reasonably be expected to result in a material violation by a Company of, or a failure on the part of a Company to comply in any material respect with, any Law, or (ii) would reasonably be expected to give rise to any obligation on the part of a Company to undertake, or to bear all or any portion of the costs of, any remedial action of any nature.

67350961.11

4.14 Legal Proceedings; Orders.

(a) Except as set forth on <u>Schedule 4.14</u>, there is no Order, Proceeding, or notification regarding a Data Breach pending, or to the Knowledge of any Owner, threatened by or against, affecting or that otherwise relates to any Company or the Business. To the Knowledge of any Owner, no event has occurred or circumstance exists that would reasonably be expected to give rise to or serve as a basis for the commencement of any Orders, Proceedings or notifications of Data Breaches against a Company.

(b) There are not now, and within the past three (3) years there have not been any material claims, actions or Proceedings pending, or threatened before any court or Governmental Authority initiated by a Company, against a Company or in respect of the Business or for which a Company would reasonably be expected to have Liability. There are no such claims, actions, Proceedings pending or, to the Knowledge of any Owner, threatened, challenging the validity of the Contemplated Transactions. No Company is now, nor has been within the past three (3) years, a party to any injunction, Order, or decree restricting the method of a Company's conduct of the Business or servicing of the Customers. Owners and the Companies have provided to Buyer copies of all notices from any Governmental Authority received by a Company in writing since January 1, 2016.

4.15 Employee Benefit Plans.

(a) <u>Schedule 4.15(a)</u> sets forth a true, complete and correct list of all Benefit Plans (1) which are currently maintained or contributed to by a Company, or (2) with respect to which a Company has any Liability or obligations to any current or former officer, employee, or service provider of a Company, or the dependents of any thereof, regardless of whether funded. With respect to each Benefit Plan, Sellers and Companies have made available to the Buyer true, accurate and complete copies of each of the following: (i) if the plan has been reduced to writing, the plan document together with all amendments thereto, (ii) if the plan has not been reduced to writing, a written summary of all material plan terms, (iii) if applicable, copies of any trust agreements, custodial agreements, insurance policies, administrative agreements and similar agreements, and investment management or investment advisory agreements, (iv) copies of any current summary plan descriptions, employee handbooks or similar employee communications, (v) in the case of any plan that is intended to be qualified under Code Section 401(a), a copy of the most recent determination, notification, or opinion letter from the IRS, and, if applicable, a copy of any pending request for such determination, (vi) in the case of any funding arrangement intended to qualify as a VEBA under Code Section 501(c)(9), a copy of the IRS letter determining that it so qualifies, (vii) in the case of any plan for which Forms 5500 are required to be filed, a copy of the two most recently filed Forms 5500, with schedules attached, (viii) actuarial valuations and reports related to any Benefit Plans with respect to the most recently completed plan years and (ix) the most recent nondiscrimination tests performed under the Code.

(b) No Company been materially liable at any time in the past six (6) years for contributions to a plan that is or has been, at any time in the past six (6) years, subject to Section 412 of the Code, Section 302 of ERISA and/or Title IV of ERISA. There is no multiemployer plan (as defined in Section 3(37) or Section 4001(a)(3) of ERISA) under which

67350961.11

any Company has any present or future Liability. In the past six (6) years no Company has sponsored or contributed to or been required to contribute to a multiemployer plan or to a multiple employer welfare arrangement (as defined in Section 3(40) of ERISA).

(c) With respect to each of the Benefit Plans, neither any Company nor any ERISA Affiliate has within the past six (6) years (i) engaged in a prohibited transaction, (ii) breached any fiduciary duty or (iii) violated any Law applicable to the Benefit Plans and related funding arrangements. Each Benefit Plan intended to be qualified under Section 401(a) of the Code has a current favorable determination letter as to its qualification or the sponsor of the Benefit Plan may reasonably rely on the IRS notification or advisory letter to the sponsor of any prototype plan or volume submitter used to document the terms of such Benefit Plan as to the tax-qualified status of such Benefit Plan, and to the Knowledge of any Owner no event has occurred which would reasonably be expected to cause any such Benefit Plan to become disqualified for purposes of Section 401(a) of the Code. Each Benefit Plan has been operated in compliance in all material respects with applicable Law, including the Code and ERISA, and in material compliance with its terms. All benefits, contributions (including employee salary deferrals) and premium payments, required to be made under the terms of any of the Benefit Plans as of the date of this Agreement have been timely paid, accrued, or, if not yet due, have been (or will be) properly reflected on the Financial Statements.

(d) All required reports, Tax Returns, documents and plan descriptions of the Benefit Plans have been timely filed with the Internal Revenue Service and the U.S. Department of Labor. No Benefit Plan has within the three (3) years prior to the date hereof been the subject of an examination or audit by a Governmental Authority or is currently the subject of an application or filing under, or is a participant in, an amnesty, voluntary compliance, self-correction, or similar program sponsored by any Governmental Authority that has not been materially resolved as of the date hereof. There are no material pending claims, lawsuits or actions relating to any Benefit Plan (other than ordinary course claims for benefits) and, to the Knowledge of any Owner, none are threatened.

(e) The consummation of the Contemplated Transactions will not accelerate the time of vesting or payment, trigger any payment or funding, or increase the amount, of compensation or benefits to any employee, officer, former employee or former officer of any Company or trigger any other material obligation pursuant to any Benefit Plan, except to the extent required under the terms of the Benefit Plan, the Code or ERISA. No Benefit Plans provide for payments that would be triggered by the consummation of the Contemplated Transactions that would subject any current or former employee or service provider of a Company to excise tax under Section 4999 of the Code, and no Company has made any payments within the past six (6) years, is not obligated to make any payments and is not a party to any agreement that would reasonably be expected to obligate it to make any payments to any current or former employee or other service provider of a Company that will not be deductible under Section 280G of the Code.

(f) Each Company has complied in all material respects with the continuation coverage provisions of COBRA and any applicable state statutes mandating health insurance

67350961.11

continuation coverage for employees. Owners and Companies have provided to Buyer a list of all current and former employees of each Company and their beneficiaries who are eligible for and/or have elected continuation coverage under COBRA or have otherwise confirmed that there are no such eligible individuals. No Benefit Plan provides for, and no written or oral agreements have been entered into by a Company promising or guaranteeing, the continuation of material medical, dental, vision, life or disability insurance coverage for any current or former employees of a Company or their beneficiaries for any period of time beyond the termination of employment (except to the extent of coverage required under COBRA or other applicable Law). Other than as required under COBRA or other applicable Law, no Benefit Plan provides material post-termination or retiree welfare benefits to any individual for any reason.

(g) Each Benefit Plan that is subject to Section 409A of the Code has been administered in material compliance with its terms and the operational and documentary requirements of Code Section 409A and all applicable regulatory guidance thereunder. No Company has any obligation to gross up, indemnify or otherwise reimburse any individual for any excise taxes, interest or penalties incurred pursuant to Code Section 409A.

(h) There has been no amendment to, or announcement by any Company relating to, any Benefit Plan that would materially increase the annual expense of maintaining such plan above the level of the expense incurred for the most recently completed fiscal year with respect to any director, officer, employee, consultant or independent contractor of a Company, as applicable, except as required under the Code or ERISA.

4.16 Employee Relations.

(a) <u>Schedule 4.16(a)</u> contains a list of all of the employees of each Company, their current salary or wage rates, bonus and other compensation, including stock options and stock grants, benefit arrangements, accrued sick days, accrued paid time off, allowed vacation days and holidays, period of service, location, department and a job title or other summary of the responsibilities of such employees. <u>Schedule 4.16(a)</u> also indicates whether such employees are part-time, full-time or on a leave of absence and the type of leave. Each Company has, at all times, complied with all applicable state and federal wage and hour Laws in all material respects, including, but not limited to, properly classifying employees as exempt or non-exempt from overtime and minimum wage requirements and properly classifying workers as independent contractors rather than employees where applicable. All employees are employees at-will, unless otherwise specified on <u>Schedule 4.16(a)</u>. No Company is delinquent in payments to any of its employees for any wages, salaries, commissions, bonuses or other direct compensation for any services performed for it or any other amounts required to be reimbursed to such employees (including accrued paid time off, accrued vacation, accrued sick leave and other benefits) or in the payment to the appropriate Governmental Authority of all required Taxes, insurance, social security and withholding thereon.

(b) Except as set forth on <u>Schedule 4.16(b)</u>, no retention payments, severance payments, change in control payments or other similar compensation or benefits are due or will be made to any employees, agencies or independent contractors of any Company in connection with the Contemplated Transactions. No Company has taken any actions prior to the Closing

67350961.11

related to the foregoing that will subject any Company or Buyer to Liability after the Closing. All employee bonuses or incentives relating to the period ending February 28, 2019 and all prior periods have been paid in full in all material respects and no bonus or related incentive plan is currently in place for any employee of any Company.

4.17 <u>Taxes</u>. For purposes of <u>Section 4.17</u> (except for <u>Section 4.17(y)</u> through <u>Section 4.17(ff)</u>) references to Company includes NTD Holdings, GH Holdco, NET Data, and GHS.

(a) All federal, state and other material Tax Returns of the Companies required to be filed by the Companies before the Closing Date (taking into account any valid extensions) have been timely filed in accordance with applicable Laws, and each such Tax Return is true, correct and complete in all material respects. All Taxes with respect to the taxable periods covered by such Tax Returns (whether or not shown on any Tax Return) before the Closing Date have been timely paid. Before the Closing Date, no Company has requested any extension of time within which to file any Tax Return which has not since been filed.

(b) The Companies have complied with the provisions of the Code relating to the withholding and payment of Taxes, including the withholding and reporting requirements under Code Sections 1441 through 1464, 3401 through 3406, and 6041 through 6049, as well as similar provisions under any other Laws, and have, within the time and in the manner prescribed by Law, withheld from employee wages and paid over to the proper Governmental Authorities all amounts required. The Companies have accurately classified all service providers as either employees or independent contractors for all Tax purposes. The Companies (i) have collected and remitted all applicable sales and/or use Taxes to the appropriate Governmental Authorities, and (ii) have obtained, in good faith, any applicable sales and/or use Tax exemption certificates.

(c) There is no written claim against a Company that was made by a Governmental Authority before the Closing Date for any Taxes which are owed by a Company and due under applicable Law, but which have not been paid in full before the Closing Date. No assessment, deficiency, or adjustment has been asserted in writing, proposed in writing, or threatened in writing by a Governmental Authority before the Closing Date with respect to any Tax Return of the Companies filed before the Closing Date. No written claim has ever been received by a Company before the Closing Date from any Governmental Authority in a jurisdiction where such Company does not file Tax Returns that it is or may be subject to Tax by that jurisdiction. No Tax audit of a Company by any Governmental Authority is currently pending or has been threatened in writing before the Closing Date. No Company has received a written notice of any proposed Tax audit, or of any Tax deficiency or adjustment before the Closing Date.

(d) No reporting position has been taken on any Tax Return that was filed before the Closing Date on behalf of a Company for a taxable period for which the statute of limitations for the assessment of Taxes with respect thereto has not expired that is the same as any Tax Return reporting position that a Governmental Authority has successfully and finally challenged before the Closing Date in the course of an examination of a Tax Return of such Company. No Company has participated in a "listed transaction," as such term is defined in

67350961.11

Section 6707A(c)(2) of the Code and Treasury Regulation Section 1.6011-4(b)(2), before the Closing.

(e) No Company is a party to or bound by any Tax sharing agreement entered into before the Closing Date or any Tax indemnity agreement, obligation or similar Contract or practice with respect to Taxes (including any agreement relating to Taxes with any Governmental Authority) entered into before the Closing Date.

(f) No Company has been a member of an affiliated group within the meaning of Section 1504(a) of the Code before the Closing Date. No Company has any Liability for Taxes of any other Person (other than the particular Company itself) under Section 1.1502-6 of the Treasury Regulations (or any similar provision of state or local Law) that arose before the Closing Date, as a transferee or successor, by Contract or otherwise.

(g) There are no Liens relating to Tax upon any properties or assets of any of the Companies immediately prior to the Closing (other than Liens relating to Taxes that are not yet due and payable or (ii) being contested in good faith and for which adequate reserves have been recorded in line items on the Financial Statements).

(h) No extensions or waivers of statutes of limitations currently in effect have been given or requested with respect to any Taxes of the Companies.

(i) The amount of the Companies' Liability for unpaid Taxes for all periods ending on or before the date of this Agreement, does not, in the aggregate, exceed the amount of accruals for Taxes (excluding reserves for deferred Taxes) reflected on the Financial Statements. The amount of the Companies' Liability for unpaid Taxes for all periods following the end of the recent period covered by the Financial Statements shall not, in the aggregate, exceed the amount of accruals for Taxes (excluding reserves for deferred Taxes) as adjusted for the passage of time in accordance with the past custom and practice of the Company

(j) <u>Schedule 4.17</u> sets forth: (i) those taxable years for which examinations by the Tax Authorities have been completed; and (ii) those taxable years for which examinations by Tax Authorities are presently being conducted

(k) All deficiencies or assessments made against either Company as a result of any examinations by any Tax Authority have been fully paid or are being contested in good faith. <u>Schedule 4.17</u> sets forth all such deficiencies or assessments.

(1) Neither Company is a party to any Proceeding by any Tax Authority. There are no pending or, to the Knowledge of any Owner, threatened Proceedings by any Tax Authority against either Company.

(m) The Owners have delivered to Buyer copies of all federal, state, local and foreign income, franchise and similar Tax Returns, examination reports, and statements of deficiencies assessed against, or agreed to by, the Companies for all Tax periods.

67350961.11

(n) There are no Liens for Taxes (other than for current Taxes not yet due and payable) upon the assets of either Company.

(o) No private letter rulings, technical advice memoranda or similar agreement or rulings have been requested, entered into or issued by any Tax Authority with respect to either Company.

(p) Neither Company will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period or portion thereof ending after the Closing Date as a result of: (i) any change in a method of accounting under Section 481 of the Code (or any comparable provision of state, local or foreign Tax Law), or use of an improper method of accounting, for a taxable period ending on or prior to the Closing Date; (ii) an installment sale or open transaction occurring prior to the Closing; (iii) a prepaid amount received on or before the Closing Date; (iv) any closing agreement under Section 7121 of the Code, or similar provision of state, local or foreign Law; (v) income from discharge of Indebtedness occurring before the Closing that was deferred pursuant to an election made under Section 108(i) of the Code or corresponding provision of state, local or foreign Tax Law; or (vi) an election by the Companies to defer any taxes attributable to a deemed repatriation of foreign assets under applicable U.S. federal, state or local Tax Law that occurred before the Closing Date.

(q) Neither Seller is a "foreign person" as that term is used in Treasury Regulations Section 1.1445-2. Neither Company is not, nor has it been, a United States real property holding corporation (as defined in Section 897(c)(2) of the Code) during the applicable period specified in Section 897(c)(1)(a) of the Code.

(r) Neither Company is, nor has it been, a party to, or a promoter of, a "reportable transaction" within the meaning of Section 6707A(c)(1) of the Code and Treasury Regulations Section 1.6011-4(b).

(s) <u>Schedule 4.17</u> sets forth for each Company each state into which such Company has, directly or indirectly, sold products or conducts business and each state for which such Company currently collects sales tax.

(t) Neither Company shall be liable for any Tax under Section 1374 of the Code in connection with the deemed sale of such Company's assets caused by the Contemplated Transactions. Neither Company has (A) acquired assets from another corporation in a transaction in which the Company's Tax basis for the acquired asset was determined, in whole or in part, by reference to the Tax basis of the acquired assets (or any other property) in the hands of the transferor or (B) acquired the stock of any subsidiary.

(u) Set forth under the heading "Tax Returns" on <u>Schedule 4.17</u> is a list of all Tax Returns filed by each Company during years 2018, 2017, and 2016.

(v) The Companies have not received or applied for any written rulings of a Governmental Authority relating to Taxes before the Closing. The Companies have not entered

27

into or countersigned a closing agreement under Section 7121 of the Code or any corresponding provision of state, local or foreign Tax Law with any Governmental Authority before the Closing that would have a continuing effect after the Closing Date.

(w) Before the Closing Date, no Company currently has, or has ever had, a permanent establishment (as defined by applicable tax treaty) or other taxable presence in any foreign country.

(x) No Company has been the "distributing company" (within the meaning of Section 355(a)(1) of the Code) or the "controlled corporation" (within the meaning of Section 355(a)(1) of the Code) (i) within the two-year period ending as of the Closing Date or (ii) in a distribution that could otherwise constitute part of a "plan" or "series of transactions" (within the meaning of Section 355(e) of the Code) in conjunction with this Agreement.

(y) Prior to the effective date of the QSub Election made for NET Data, NET Data was, at all times since its formation, a validly electing and qualifying S corporation within the meaning of Section 1361(a) of the Code for U.S. federal and state income tax purposes.

(z) Since its formation NTD Holdings has been a validly electing and qualifying S corporation within the meaning of Section 1361(a) of the Code for U.S. federal and state income tax purposes and no election, action, or failure to take any action that would terminate such status has occurred or will occur prior to the Closing.

(aa) Beginning on the effective date of the QSub Election made for NET Data and until the Disregarded Entity Conversion for NET Data, NET Data was a QSub for U.S. federal and state income tax purposes.

(bb) As of the effective date of the Disregarded Entity Conversion for NET Data, NET Data was a Disregarded Entity for U.S. federal and state income tax purposes and no election, action, or failure to take any action that would terminate such status has occurred or will occur prior to the Closing.

(cc) Prior to the effective date of the QSub Election made for GHS, GHS was, at all times since its formation, a validly electing and qualifying S corporation within the meaning of Section 1361(a) of the Code for U.S. federal and state income tax purposes.

(dd) Since its formation GH Holdco has been a validly electing and qualifying S corporation within the meaning of Section 1361(a) of the Code for U.S. federal and state income tax purposes and no election, action, or failure to take any action that would terminate such status has occurred or will occur prior to the Closing.

(ee) Beginning on the effective date of the QSub Election made for GHS and until the Disregarded Entity Conversion for GHS, GHS was a QSub for U.S. federal and state income tax purposes.

67350961.11

(ff) As of the effective date of the Disregarded Entity Conversion for GHS, GHS was a Disregarded Entity for U.S. federal and state income tax purposes and no election, action, or failure to take any action that would terminate such status has occurred or will occur prior to the Closing.

4.18 Environmental Matters.

(a) Each Company is currently in compliance in all material respects with all Environmental Laws which compliance includes, but is not limited to, the possession by each Company of all Licenses and other governmental authorization required under applicable Environmental Laws and in compliance in all material respects with the terms and conditions thereof to operate the Business as currently operated.

(b) Since the date of each Company's leasing of the Real Property, there have been no actions, activities, circumstances, conditions, events or incidents that could reasonably be expected to form the basis of any material Environmental Claim against any Company with respect to the Real Property, and no Owner has Knowledge of any such actions, activities, circumstances, conditions, events or incidents prior to the Companies' ownership or leasing of the Real Property or assets.

4.19 <u>Affiliate Transactions</u>. Except as set forth on <u>Schedule 4.19</u>, no Company has been a party to any material business arrangement or relationship with an Owner within the past twelve (12) months, and no Owner or any of an Owner's affiliates own any asset, tangible or intangible, that is used by any Company. Except as set forth on <u>Schedule 4.19</u>, no Owner or any of an Owner's affiliates own, or in the last twelve (12) months has owned, of record or as a beneficial owner, any financial or profit interest in any Person that has had business dealings or a material financial interest in any transaction with any Company. Except as set forth on <u>Schedule 4.19</u>, since December 31, 2018 no cash distribution has been made by a Company to any Owner.

4.20 Insurance. Schedule 4.20 sets forth an accurate description of each insurance policy currently utilized and/or maintained by any of the Companies with respect to its properties, assets and Business. Such insurance policies are of the types, in the amounts and covering such risks as are customarily carried by Persons conducting a business similar to the Business. All of such insurance policies are, and immediately following the Closing will continue to be, legal, valid, binding and enforceable and in full force and effect, and the Companies are not in material breach or default (or with the giving of notice or lapse of time or both, would be in breach or default) with respect to their obligations under such insurance policies (including with respect to payment of premiums). Except as set forth on <u>Schedule 4.20</u>, the Companies maintain all insurance policies required to be maintained by them pursuant to the terms and conditions of any active Material Contract. As of the date hereof, no Company has received (a) notice of cancellation or notice of failure to renew of any such insurance policy or refusal of coverage thereunder, (b) notice that any issuer of such policy has filed for protection under applicable bankruptcy Laws or is otherwise in the process of liquidating or has been liquidated, (c) notice of any material adjustment in the amount of the premiums payable with respect to any such insurance policy, or (d) any other notice that such policies are no longer in full force or

67350961.11

effect or that the issuer of any such policy is no longer willing or able to perform its obligations thereunder. All premiums payable with respect to such policies that are due and payable have been paid. Except as set forth on <u>Schedule 4.20</u>, there is no pending claim under any such insurance policy.

4.21 Customer Arrangements.

(a) <u>Agreements with Customers.</u>

(i) Attached to <u>Schedule 4.21(a)</u> are true and correct copies of the forms upon which (A) any agreements with or applications from Customers or (B) any currently effective Customer Agreements are based. The ten (10) largest Customers, as measured by revenue generated during the twelve (12) months prior to the Balance Sheet Date, are identified as such on <u>Schedule 4.21(a)(i)</u>.

(ii) With respect to the Customers, whether as a result of the Contemplated Transactions or otherwise, none of the Customer Agreements obligate or will obligate any Company (by the terms of any of such Contracts, or at the option of the other party to such Contracts) to (A) make a lump sum payment in lieu of any future stream of revenue or otherwise, (B) acquire or assume any asset or Liability, (C) offer a right of first refusal or similar preferential right in favor of such Customer or any Referral Source or (D) pay ongoing residuals, commissions or fees to any third party.

(b) <u>Ownership of Customer Agreements</u>. As of the Closing, each Company has the right, without giving notice to or receiving consent (except the Required Consents) from the Customers or any other third party, to engage in the Contemplated Transactions. No transfer, assignment or other fee will be due to any Customer or Processor in connection with the Contemplated Transactions.

(c) <u>Material Customers</u>. <u>Schedule 4.21(c)</u> provides a list of each Customer that on an annual basis for calendar year 2016, 2017 or 2018 was among the top twenty (20) Customers of each Company in revenue (to such Company) (a "<u>Material Customer</u>"). There is no existing dispute between any Company and a Material Customer and, to the Knowledge of any Owner, no Material Customer has provided any Company written notice that it intends to terminate its Customer Agreement or materially reduce its annual credit/debit card volume whether as a result of the consummation of the Contemplated Transactions or otherwise. There are no Material Customers that receive services from any Company, or in respect of whom any Company receives payments, that are not a party to a written Customer Agreement.

(d) <u>Customer Compliance with Rules</u>. To the Knowledge of any Owner, no Company has received notice of any Customer's material breach of the Rules.

(e) <u>Customer Reserves</u>. Except as set forth on <u>Schedule 4.21(e)</u>, no Company maintains any Customer reserves, held funds or Customer liability accounts.

67350961.11

(f) <u>Customer Status</u>. To the Knowledge of any Owner, no Company has received notice that any Customer:

(i) is a "Payment Facilitator" as such term is defined in the Rules;

(ii) participates in or is required to participate in any "Risk Monitoring Service," "Excessive Chargeback Monitoring," or similar programs or "Chargeback Monitored Customers" as defined in the Rules;

(iii) has been fined by (or has fines pending from) any Payment Network, or

(iv) has been placed on the "MATCH" (Customer Alert to Control High-Risk) File, "CTMF" (Combined Terminated Customer File) or any similar list or database.

(g) <u>Customer Underwriting Guidelines</u>. Owners and Companies have provided to Buyer a copy of each Company's merchant underwriting guidelines. To the actual knowledge of the Owners, each of the Customers is in compliance, in all material respects, with each Company's merchant underwriting guidelines and is in compliance, in all material respects, with the following Laws: the US Patriot Act, the Bank Secrecy Act, the Anti-Money Laundering Act, the Consumer Financial Protection Bureau's guidance on Unfair, Deceptive or Abusive Acts, the Fair Credit Reporting Act, the Telephone Consumer Protection Act, the Restore Online Shopper's Confidence Act and the Unlawful Internet Gambling Enforcement Act of 2006. In the three (3) years prior to the Closing Date, no Company has received any subpoena or other written inquiry from any Governmental Authority regarding any Customer.

4.22 <u>Inventory</u>. Any inventory of the Companies, whether or not reflected in the Financial Statements, consists of a quality and quantity usable and salable in the ordinary course of business consistent with past practice, except for obsolete, damaged, defective or slow-moving items that have been written off or written down to fair market value or for which adequate reserves have been established. Any inventory of the Companies is owned by the Companies free and clear of all Liens, and no inventory is held on a consignment basis. The quantities of each item of inventory (whether raw materials, work-in-process or finished goods) are not excessive, but are reasonable in the present circumstances of the Companies.

4.23 Privacy.

(a) To the Knowledge of Company, each Company is and has been in compliance with: (i) all applicable Privacy Laws, (ii) all of such Company's policies regarding privacy and (iii) all contractual commitments that such Company has entered into with respect to Personal Information.

(b) Each Company has commercially reasonable safeguards in place to protect any Personal Information in its possession or control from unauthorized access, including by its employees, contractors and consultants.

67350961.11

(c) To the Knowledge of Company, no Company has made any illegal or unauthorized use of Personal Information.

(d) The transfer, if any, of Personal Information in connection with the Contemplated Transactions will not violate any Privacy Laws. No Company is subject to any contractual requirements, privacy policies or other legal obligations that, following the Closing, would prohibit a Company, Buyer or any of its affiliates from receiving or using Personal Information in the manner in which a Company receives and uses such Personal Information prior to the Closing.

(e) No Company has received any notice of any claims against a Company or been charged with the violation of any Privacy Laws. To the Knowledge of any Owner, no Company has been nor is currently under investigation with respect to any violation of any Privacy Laws or applicable privacy policies.

(f) There have been no Data Breaches involving any Company and/or any Personal Information in the possession of any Company.

4.24 Referral Sources.

(a) <u>Referral Source Agreements</u>. Attached as <u>Schedule 4.24(a)</u> are true and correct copies of the forms upon which any currently effective Referral Source Agreements are based.

(b) <u>Referral Source List</u>. <u>Schedule 4.24(b)</u> (i) sets forth the name of each Referral Source who has received payments from any Company in the last three years and (ii) sets forth a complete list of agreements between any Company and the Referral Sources. Next to each Referral Source name, <u>Schedule 4.24(b)</u> sets forth (1) a list of each Referral Source that on an annual basis for calendar year 2016, 2017 or 2018 was among the top twenty (20) Referral Source of each Company in revenue (to such Company) (a "<u>Material Referral Source</u>") and (2) for calendar year 2016, 2017 and 2018, the size of the payments made from each Company to each Referral Source, the size of the payments made from each Referral Source to such Company, and the aggregate volume of payments made by Customers of each Referral Source. There are no Material Referral Sources that are not a party to a written Referral Source Agreement.

(c) <u>Disputes</u>. Except for as described on <u>Schedule 4.24(c)</u>, there is no existing dispute between any Company and any Referral Source and, to the Knowledge of any Owner, no Referral Source intends to terminate its agreement with any Company or materially reduce its referral volume as a result of the consummation of the Contemplated Transactions or otherwise.

4.25 Data Security.

(a) To the Knowledge of any Owner, no Person has gained unauthorized access to the Platform, any computers, computer software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines, or other information

67350961.11

technology equipment, and associated documentation used by or on behalf of any Company or its affiliates or any Processor or service provider of any Company or its affiliates or any data stored thereon (including any Customer data, Personal Information, or cardholder data) (a "<u>Data Breach</u>"). None of any Company, its affiliates or, to the Knowledge of any Owner, any Processor, Referral Source or Customer, has received a "common point of purchase", "point of compromise" or similar notice, letter or inquiry. No third-party network penetration test, audit or review of any Company's external and internal information technology network has identified any material risks relating to the security of any Company's network. Each Company has been at all times a compliant service provider of each applicable Card Brand.

(b) No Company is required to be in compliance with the Payment Card Industry Data Security Standards because it does not store, transmit or process "cardholder data" or "sensitive authentication data" (as such terms are defined in the Payment Card Industry Data Security Standards ("<u>PCI-DSS</u>")).

(c) Each third-party vendor of any component included in any Company Product that stores, transmits or processes "cardholder data" or "sensitive authentication data" (as such terms are defined in the PCI-DSS) (a "<u>Third-Party PCI-DSS</u> <u>Component</u>"), including any "white label" payment gateway component, is currently in compliance with the PCI-DSS and is a PCI Level 1 validated service provider. Each Third-Party PCI-DSS Component that is required to be validated under the Payment Application Data Security Standards is so validated.

(d) Each Company is and has been in compliance with (i) all applicable Laws regarding data security, (ii) all of such Company's policies regarding data security and (iii) all contractual commitments of such Company with respect to data security.

(e) All Company Products that are required to be validated under the Payment Application Data Security Standards are so validated.

4.26 <u>Cryptocurrencies</u>. No Company has ever (a) owned any Bitcoin, Ether or other cryptocurrencies ("<u>Cryptocurrencies</u>"), (b) processed Cryptocurrencies on behalf of any Customers or (c) been a party to any Contracts with any processor, exchange or wallet of or for Cryptocurrencies.

4.27 <u>Debt Collection</u>. No Company provides, and has not provided or marketed, services to any Customer that impose convenience fees or similar fees where state law prohibits the collection of such convenience or similar fees. The Companies are not engaged and have not in the five (5) years prior to the Closing Date been engaged in any debt collection activities as defined under the U.S. Fair Debt Collection Practices Act or as contemplated under the Texas Fair Debt Collection Practices Act. The Companies are, and for the last five (5) years have been at all times, in compliance with all Laws related to collections of fines imposed by a Governmental Authority, including the collection of fees related thereto and have held all necessary licenses, permits and registrations required under applicable Laws with respect to the Companies' respective fine collection activities.

67350961.11

4.28 <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in <u>Article III</u> and this <u>Article IV</u> (including the related portions of the Disclosure Schedules), no Owner has made or makes any other express or implied representation or warranty, either written or oral, on behalf of Owner, Sellers or the Companies.

ARTICLE V - REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to Sellers and Owners as of the Closing Date as follows:

5.1 <u>Limited Liability Company Status</u>. Buyer (a) is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware, and (b) has all requisite limited liability company power and authority to carry on its business as now and heretofore conducted and to own, operate and lease its properties.

5.2 Limited Liability Company Authorization; Binding Effect. Buyer has full limited liability company power and authority to execute and deliver the Transaction Documents and to perform its obligations hereunder and thereunder. The Transaction Documents and performance and consummation of the Contemplated Transactions are and have been approved by all requisite limited liability company action of Buyer. No other limited liability company action or Proceedings on the part of Buyer are necessary to approve or authorize the execution and delivery of the Transaction Documents and the consummation of the Contemplated Transaction Documents and the consummation of the Contemplated Transactions. The Transaction Documents have been duly executed and delivered by Buyer and, assuming due authorization, execution and delivery of the Transaction Documents by Sellers and Owners, each Transaction Document is the valid and legally binding obligation of Buyer, enforceable in accordance with its terms, subject to applicable bankruptcy, reorganization, insolvency, moratorium and other Laws affecting creditor's rights generally from time to time in effect and limitations on enforcement of equitable remedies.

5.3 <u>No Conflict</u>. Buyer's execution, delivery and performance of the Transaction Documents and consummation of the Contemplated Transactions and compliance with any of the provisions thereof will not (a) violate any Law to which Buyer may be subject, (b) conflict with or result in a breach of any provision of the limited liability company operating agreement of Buyer, (c) require any consent, approval or authorization of, or notice to, or declaration, filing or registration with, any Governmental Authority that is required by Law or the regulations of any Governmental Authority, (d) violate any Order of any Governmental Authority to which Buyer may be subject, or (e) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice or lapse of time or both would constitute a default) under, or result in the termination of or in a right of termination or cancellation of, or accelerate the performance required by, or result in the creation of any Lien upon any of the properties of Buyer or result in being declared void, voidable, without further binding effect, or subject to amendment or modification any of the terms, conditions or provisions of, any contract, license, franchise, permit, or other material instrument or commitment or obligation to which Buyer may be bound or affected.

5.4 <u>Brokers and Finders</u>. Neither Buyer nor its agents has engaged any finder or broker or incurred any obligation or Liability, contingent or otherwise, for brokerage or finders'

67350961.11

fees or agents' commissions in connection with this Agreement and the Contemplated Transactions.

5.5 <u>Certain Proceedings</u>. There is no pending Proceeding that has been commenced against Buyer that challenges, or may have the effect of preventing, delaying, making illegal or otherwise interfering with, any of the transactions contemplated hereby. To Buyer's Knowledge, no such Proceeding has been threatened against Buyer.

5.6 Inspection; No Other Representations; No Reliance.

(a) Buyer is an informed and sophisticated Person, and has engaged expert advisors experienced in the evaluation and acquisition of companies such as the Companies as contemplated hereunder. Buyer acknowledges and agrees that it is relying exclusively on the representations, warranties and covenants of Sellers and Owners set forth in this Agreement, and its own examination and investigation of Sellers, Owners and the Companies, and it is not relying on any other statements or documents.

(b) Without limiting the generality of the foregoing, Buyer acknowledges that, except as may be set forth in the representations and warranties contained in <u>Article III</u> and <u>Article IV</u> (i) none of the Companies or Owners or any of their respective affiliates makes any representation or warranty with respect to (A) any projections, estimates or budgets delivered to or made available to Buyer of future results of operations (or any component thereof) or future financial condition (or any component thereof) of the Companies or the future business and operations of the Company or (B) any other information or documents made available to Buyer or any of its affiliates with respect to the Companies or their businesses, assets, liabilities or operations, except as expressly set forth in this Agreement, and (ii) Buyer has not relied and will not rely upon any of the information described in sub-clauses (A) and (B) of clause (i) above in executing, delivering and performing this Agreement and the other agreements, documents, instruments or certificates contemplated hereby and the transactions or warranties set forth in this Agreement in negotiating, discussing, executing, delivering and performing this Agreements, instruments or certificates contemplated hereby and the other agreements, documents, instruments or certificates contemplated hereby and the other agreements, instruments or certificates contemplated hereby and the other agreements, instruments or certificates contemplated hereby and the other agreements, instruments or certificates contemplated hereby and the other agreements, instruments or certificates contemplated hereby and the other agreements, instruments or certificates contemplated hereby and the other agreements, instruments or certificates contemplated hereby and the other agreements, instruments or certificates contemplated hereby and thereby.

5.7 <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in this <u>Article V</u>, Buyer has neither made nor makes any other express or implied representation or warranty, either written or oral, with respect to Buyer or the Contemplated Transactions.

ARTICLE VI - COVENANTS OF SELLERS AND OWNERS

6.1 <u>Notices and Consents</u>. To the extent that Sellers have not obtained all necessary Licenses, consents, waivers or other authorizations or approvals or Required Consents from any Governmental Authority or any private third-party as of the Closing and the Parties nonetheless elect to close, then upon request by Buyer, Sellers shall use their commercially reasonable efforts to: (a) obtain such consents as soon as practicable post-Closing, (b) cooperate with Buyer in any

67350961.11

reasonable and lawful arrangements under which Buyer would obtain the benefit of the matter concerned; and (c) enforce for the account of Buyer any rights of Sellers arising from the matter concerned, provided, however, that Sellers shall not be required to make any payments or incur any additional Liabilities in connection with such efforts.

6.2 Tax Matters.

(a) <u>Pre-Closing Tax Returns</u>.

(i) Sellers will prepare and file on a timely basis, all federal, state, and local income Tax Returns with respect to the Companies for taxable periods ending on or before the Closing Date ("<u>Pre-Closing Tax Periods</u>") that are not filed on or before the Closing Date. All such Tax Returns shall be filed in accordance with the past practices of the Companies unless contrary to applicable Law. Sellers shall include on the Tax Returns for the Pre-Closing Tax Periods the income of the Companies for all applicable Pre-Closing Tax Periods. Sellers (i) will submit all such Tax Returns (including all attachments thereto) to Buyer for its review at least thirty (30) days prior to filing and accept all comments of Buyer that are reasonable, and (ii) will, promptly after filing, forward to Buyer a true, correct and complete copy of such filed Tax Returns (including all attachments thereto) and proof of payment of the subject Taxes. Sellers shall pay all fees and expenses associated with preparing the income Tax Returns for the Pre-Closing Tax Periods.

(ii) Buyer will cause the Companies to prepare and file on a timely basis, (a) all non-income Tax Returns for the Pre-Closing Tax Period and (b) all Tax Returns for Tax periods beginning before and ending after the Closing Date ("<u>Straddle Tax Period</u>").

(b) Payment of Taxes.

(i) To the extent that Taxes of the Companies for all Pre-Closing Tax Periods and portions of any Straddle Tax Period ending on the Closing Date (including all such Taxes payable with respect to Tax Returns filed under this <u>Section 6.2</u> and any Taxes assessed after the Closing with respect to Pre-Closing Tax Periods and portions of any Straddle Tax Period ending on the Closing Date) are expressly accrued or reserved for as Current Liabilities in line items on the Closing Statement and taken into account in determining the Net Working Capital Calculation, Buyer will pay or cause to be paid such Taxes. To the extent such Taxes for Pre-Closing Tax Periods and portions of any Straddle Tax Period ending on the Closing Date are not so reflected as Current Liabilities on the Closing Statement and not taken into account in determining the Net Working Calculation, Sellers will pay all such Taxes, including all Taxes incurred by the Companies including any Taxes from the payments under any employee sale bonuses, change of control agreements or any other payments that are treated as compensation for federal, state or local income Tax purposes, including the Companies' portion of such Taxes. Taxes that are payable with respect to a Straddle Tax Period will be allocated to the portion of the period that ends on the Closing Date in accordance with <u>Section 6.2(c)</u>.

(ii) In the case of Tax Returns filed by Buyer under this <u>Section 6.2</u> and as to which Buyer expects payment from Sellers, Buyer shall deliver the pertinent Tax

67350961.11

Return to the Representative and inform the Representative of any amounts due from Sellers at least ten (10) days prior to the due date of the pertinent Tax Return, and Sellers will pay such amounts to Buyer in immediately available funds at least five (5) days prior to the due date of the Tax Return. In the case of any Tax Returns filed by Sellers, Sellers will pay the amount of Taxes due with respect to such Tax Returns and provide proof of payment to Buyer no later than five (5) days after the applicable payment is made to the applicable Governmental Authority.

(c) <u>Tax Apportionment</u>. In the case of Taxes that are payable with respect to a Straddle Tax Period, the portion of any such Tax that is allocable to the portion of the Straddle Tax Period ending on the Closing Date will be (i) in the case of Taxes that are either (x) based upon or related to income or receipts or (y) imposed in connection with any sale or other transfer or assignment of property (real or personal, tangible or intangible), deemed equal to the amount which would be payable if the taxable period ended as of the close of business on the Closing Date; *provided, however*, that all exemptions, allowances, or deductions for the Straddle Tax Period which are calculated on an annual basis (including depreciation and amortization deductions) shall be allocated in proportion to the number of days in each period; and (ii) in the case of Taxes imposed on a periodic basis with respect to the assets of the Companies, or otherwise measured by the level of any item, deemed to be the amount of such Taxes for the entire period (or, in the case of such Taxes determined on an arrears basis, the amount of such Taxes for the immediately preceding period), <u>multiplied</u> by a fraction the numerator of which is the number of calendar days in the period.

(d) <u>Transactional Taxes</u>. Notwithstanding any other provision of this Agreement, all transfer, documentary, recording, notarial, sales, use, registration, stamp and other similar Taxes or fees imposed by any Governmental Authority in connection with the Contemplated Transactions will be borne by Sellers. Sellers will, at their own expense, file all necessary Tax Returns and other documentation with respect to all such Taxes and, if required by applicable Law, Buyer will join in the execution of any such Tax Returns or other documentation.

(e) Audits and Claims.

(i) Sellers shall have the right (at Sellers' sole risk, cost and expense) to control the conduct of any audit, examination, investigation or administrative, court or other Proceeding related to income Taxes for any Pre-Closing Tax Period ("<u>Income Tax Proceeding</u>"), <u>provided</u>, <u>however</u>, that such Seller or Sellers acknowledge and agree in writing to assume all liability for indemnification of all Tax, interest, penalties, and other costs associated with such Income Tax Proceeding. Buyer shall be entitled to attend and participate in any such Income Tax Proceeding at its sole cost and expense and shall be permitted to receive copies of any pleadings, correspondence, and other documents as Buyer may reasonably request. Buyer shall have the right to consent to any settlement related to an Income Tax Proceeding if such settlement could increase the Taxes of the Companies for any period (or portion thereof) beginning on or after the Closing Date.

67350961.11

(ii) Except for Income Tax Proceedings, Buyer shall have the right (at Buyer's cost and expense) to control the conduct of any other audit, examination, investigation or administrative, court or other Proceeding ("<u>Tax Proceedings</u>"). The Representative shall be entitled to attend and participate in any such Tax Proceedings at its sole cost and expense if such Tax Proceedings could result in a claim for indemnification for any Tax against Sellers hereunder.

(iii) If Buyer or Sellers receive any written or oral communication with respect to a Tax claim for which the other Party has potential liability under the terms of this Agreement, then such Party shall promptly notify the other Party (which for Sellers shall be the Representative) hereto in writing of the existence of such Tax claim.

(f) Cooperation on Tax Matters.

(i) Buyer, the Companies, the Representative and Sellers shall cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the filing of Tax Returns and any Income Tax Proceedings or Tax Proceeding. Such cooperation shall include the retention and (upon the other Party's request) the provision of records and information reasonably relevant to any such audit, litigation, or other Proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. The Companies, the Representative, Sellers and Buyer shall (i) retain all Books and Records with respect to Tax matters pertinent to the Business relating to any taxable period beginning before the Closing Date until expiration of the statute of limitations (and, to the extent notified by Buyer or Sellers, any extensions thereof) of the respective taxable periods, and to abide by all record retention agreements entered into with any Governmental Authority, and (ii) give the other Party (which for Sellers, shall be the Representative) reasonable written notice prior to transferring, destroying, or discarding any such books and records and, if the other Party (which for Sellers, shall be the Representative) so requests, allow such Party to take possession of such books and records.

(ii) Buyer and Sellers and the Representative further agree, upon request, to use commercially reasonable efforts to obtain any certificate or other document from any Governmental Authority or any other Person as may be necessary to mitigate, reduce, or eliminate any Tax that could be imposed (including with respect to the transactions contemplated hereby).

(g) <u>Withholding</u>. If Buyer provides reasonable advance notice and satisfactory justification, Buyer will be entitled to deduct and withhold from any amount payable pursuant to this Agreement such amounts as Buyer shall determine they are required to deduct and withhold with respect to the making of such payment under the Code or any other provision of applicable Law. To the extent that amounts are so withheld by Buyer, such withheld amounts will be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding were made.

(h) Reserved.

67350961.11

(i) <u>Allocation of Purchase Price</u>. Because each Company is a Disregarded Entity, the Parties agree that for U.S. federal and state income tax purposes, the sale and purchase of the Equity Interests shall be treated as: (i) a taxable sale of assets by NTD Holdings and GH Holdco, respectively, in exchange for their respective shares of the Purchase Price (and any other amount treated as an amount realized on such sale); and (ii) an acquisition of assets by Buyer in exchange for the Purchase Price paid under this Agreement and any other amount that is treated as part of the consideration for Tax purposes. The Parties agree that the allocation of the Purchase Price among the assets of each Company for all purposes (including financial accounting and Tax purposes) shall be made by Buyer and shall be set forth in a statement prepared by Buyer in accordance with the Code (the "<u>Tax Allocation Statement</u>"). The Parties shall cooperate in the preparation of the Tax Allocation Statement which shall be filed as required by Law. The Parties agree to file their respective IRS Form 8594 and all federal, state and local Tax Returns (including amended returns and claims for refund) in accordance with the Tax Allocation Statement. Buyer shall deliver the final version of the Tax Allocation Statement to Representative no later than 30 days following the date the Final Adjustment Calculation is determined pursuant to <u>Section 2.3</u>.

(j) Gross Up Payment.

(i) Notwithstanding anything in this Agreement to the contrary, in the event Buyer, as reflected in the Tax Allocation Statement, allocates:

(A) more than \$70,000 of the Purchase Price to the restrictive covenants described in <u>Section 6.3</u> (such amount in excess of \$70,000, the "<u>Restrictive Covenant Excess Allocation</u>");

(B) more than \$320,000 to Class V assets (as such class allocation concept is described in Treas. Regs. 1.1060-1 and 1.338-6, 1.338-7) (such amount in excess of \$320,000, the "<u>Class V Excess Allocation</u>"); or

(C) more than \$3,500,000 to Class VI self-created software assets (as such class allocation concept is described in Treas. Regs. 1.1060-1 and 1.338-6, 1.338-7) (such amount in excess of \$3,500,000, the "<u>Class VI</u> <u>Excess Allocation</u>"),

then Buyer shall, within fifteen (15) days after delivery of the final Tax Allocation Statement to Representative pursuant to <u>Section 6.2(i)</u>, make a one-time aggregate cash payment to Sellers (the "<u>Gross Up Payment</u>") in an amount equal to the quotient of (A) the product of (i) the sum of the Restrictive Covenant Excess Allocation, the Class V Excess Allocation and the Class VI Excess Allocation and (ii) 13.2% divided by (B) 76.20%. For example, if the Restrictive Covenant Excess Allocation is \$10,000, the Class V Excess Allocation is \$40,000 and the Class VI Excess Allocation is \$50,000 then the Gross Up Payment would be \$17,322.83. The Gross Up Payment shall be added to goodwill on the Tax Allocation Statement.

6.3 Non-competition; Non-solicitation.

67350961.11

(a) During the period beginning on the Closing Date and ending three (3) years from the Closing Date, none of the Owners will, or permit any of their affiliates (including any Seller) to, without the prior written consent of Buyer:

(i) work with, provide services to, or own an interest in any business that competes directly or indirectly with the Business or the i3V Business, provided, that, the foregoing agreement shall not be deemed to restrict the ownership of an Owner of up to five percent (5%) of any class of the outstanding capital stock of any corporation conducting a business similar to the Business or the i3V Business which is traded on a national securities exchange;

(ii) hire, attempt to hire, solicit, induce, or attempt to solicit or induce any employee or independent contractor of any Company, i3V or any of its affiliates to leave any Company's, i3V's or any of its affiliates' employment or to terminate his or her or its contractual and/or business relationship with any Company, i3V or any of its affiliates; provided, however, that general advertising not targeting any Company's, i3V's or any of its affiliates' employees shall not be restricted by this Agreement;

(iii) call on, divert or solicit, any of the Customers;

(iv) work with or provide services related to the Business to any Customer or other customer of any Company, i3V, Buyer or any of their respective affiliates; or

(v) encourage or persuade any Customer or other customer of any Company, i3V, Buyer or any of their respective affiliates not to enter into an agreement or to terminate an agreement with any Company, i3V, Buyer or any of their respective affiliates or to obtain similar services from a competitor of any Company, i3V, Buyer or any of their respective affiliates.

(b) <u>Non-Disparagement</u>. For five (5) years after the Closing Date, the Parties agree that they shall not, and shall make reasonable efforts to ensure that their respective affiliates, managers, employees, and officers do not make any written or verbal statements, whether in commercial or noncommercial speech, disparaging or criticizing in any way the other Party, or such other Party's affiliates, equity owners, members, managers, employees, directors, trustees or officers (collectively, the "<u>Protected Party</u>"), or any products or services offered by the Protected Party. The foregoing shall not be interpreted to prevent a Party from making in good faith any statement or assertion regarding a Protected Party (i) to the extent reasonably necessary to comply with applicable Law or an Order issued by a Governmental Authority of competent jurisdiction, (ii) for the sole purpose of enforcing a Party's rights under the Transaction Documents, (iii) for the purpose of defending against, and only to the extent reasonably necessary to defend against, claims made by or against a Party under any of the Transaction Documents, or (iv) in connection with protecting or preserving its business, including under contracts used in the operation of its business.

(c) The consideration for this <u>Section 6.3</u> is included in the Purchase Price. Sellers and Owners specifically acknowledge and agree that the remedy at law for any breach of

67350961.11

this <u>Section 6.3</u> will be inadequate and that Buyer, in addition to any other relief available to it, may be entitled to temporary and permanent injunctive relief.

6.4 <u>Owners' Release</u>. Each Seller and Owner, on its own behalf, and on behalf of each of its affiliates irrevocably, unconditionally, voluntarily, knowingly, fully, finally and completely forever releases and discharges each of Buyer, its affiliates (including i3V) and the Companies and their respective successors, assigns and predecessors and each of their respective directors, managers, officers, employees, shareholders, and members (the "<u>Released Parties</u>") from, against and with respect to any and all actions, accounts, agreements, bonds, causes of action, complaints, charges, claims, contracts, controversies, costs, covenants, damages, demands, debts, defenses, duties, expenses, executions, fees, injuries, interest, judgments, liabilities, losses, obligations, penalties, promises, reimbursements, remedies, suits, sums of money, torts, and undertakings of whatever kind or character, whether in law, equity or otherwise, or direct or indirect (including Damages), that such Seller or Owner now has, or may hereafter have or acquire, against any of the Released Parties (collectively, the "<u>Released Claims</u>"); provided, however, that nothing contained herein will operate to release any obligations of any of the Released Parties or any rights of any Seller or Owner arising under this Agreement or under any Transaction Document. Each Seller and Owner, on behalf of itself and its affiliates, irrevocably agrees to refrain from directly or indirectly asserting any claim or demand or commencing (or causing to be commenced) any Proceeding of any kind, in any court or before any tribunal, against any Released Party based upon any Released Claim.

ARTICLE VII - COVENANTS OF BUYER, SELLERS AND OWNERS

7.1 Post-Closing Access to Information.

(a) Sellers and Buyer acknowledge that subsequent to the Closing each Party may need access to information or documents in the control or possession of the other Party for the purposes of concluding the Contemplated Transactions, audits, compliance with Laws and governmental requirements, and the prosecution or defense of third-party claims. Accordingly, Sellers and Buyer agree that until the later of the four (4) year anniversary of the Closing Date or the expiration of any applicable statute of limitations pertaining to Tax matters, to the extent permitted by Law, each will make reasonably available to the other's agents, independent auditors and/or governmental agencies upon written request and at the expense of the requesting Party such documents and information as may be available for periods prior and subsequent to Closing to the extent necessary to facilitate concluding the Contemplated Transactions, audits, compliance with Laws and governmental requirements and regulations and the prosecution or defense of third-party claims.

(b) Upon written request, each of the Parties shall cooperate with the other Parties in good faith, at the requesting Party's expense, in furnishing information, testimony and other assistance in connection with any actions, Proceedings, arrangements, or disputes involving any of the Parties (other than in a dispute among such parties or entities) and based upon contracts, arrangements or acts of the Company or any Party which were in effect or occurred prior to the Closing. Buyer shall cause to be provided any information or documents reasonably requested by Sellers in connection with Tax or other disputes, settlements, investigations,

67350961.11

Proceedings or other matters in respect of any period ending at or prior to the Closing. The Party requesting documents or information pursuant to this <u>Section 7.1</u> shall pay all fees and expenses paid to unaffiliated third parties by the Party providing such documents or information in connection with providing such information or documents.

7.2 <u>Further Assurances and Cooperation</u>. Sellers and Owners shall, at any time and from time to time at and after the Closing, upon the request of Buyer, take any and all steps reasonably necessary and under any Seller's or Owner's control to place Buyer in possession and operating control of the Company and the Equity Interests and will do, execute, acknowledge and deliver, or will cause to be done, executed, acknowledged and delivered, all such further acts, deeds, assignments, transfers, conveyances, powers of attorney and assurances as may be reasonably required to transfer and confirm more effectively to Buyer or to its successors or assigns, or to reduce to possession, any or all of such Equity Interests and to carry out the purposes and intent of this Agreement. From and after the date of this Agreement, upon the request of any Party, the other Parties shall furnish such further information, execute and deliver such schedules, instruments, documents or other writings and take such actions as may be reasonably necessary or desirable to confirm and carry out and to fully effectuate the intent and purposes of this Agreement.

7.3 <u>Confidentiality</u>. Each Party agrees that the existence of this Agreement (including any reference to the Contemplated Transactions generally), the existence of any dispute (if any) between the Parties under this Agreement, and the terms and conditions hereof and the other Transaction Documents shall not be disclosed to any third party without prior written consent of the other Parties, and no Party, nor any representative thereof shall distribute or provide access to this Agreement or the other Transaction Documents, or the contents or any part thereof, to any third party, except to such Party's legal and financial advisors, lenders and accountants and in any case as required by Law or accounting practice or to allow attorneys to enforce or interpret this Agreement, or by Buyer or its affiliates to fulfill i3V's obligations under any Securities and Exchange Commission or Nasdaq requirements, whether such disclosure is made publicly or privately.

ARTICLE VIII - INDEMNIFICATION

8.1 <u>Survival</u>. The representations and warranties made by each Party in this Agreement shall survive the Closing Date and shall continue in full force and effect for a period of eighteen (18) months thereafter; provided, however, that (a) each of the Fundamental Representations shall survive until the maximum allowable statute of limitations applicable to Actions based on written contracts under applicable Law, and (b) each of the Special Representations shall survive indefinitely or for the period stated therein. In each case, the period from the Closing Date until the last date on which a representation, warranty, covenant or other obligation survives is referred to herein as the "<u>Survival Period</u>".

8.2 <u>Indemnification by Owners</u>. Subject to the other provisions of this <u>Article VIII</u>, each Owner agrees to (on a several and not joint basis in accordance with <u>Section 8.4(d)</u>)

indemnify, defend and hold harmless Buyer Indemnified Parties from and against and shall pay to Buyer Indemnified Parties the amount of, or reimburse Buyer Indemnified Parties for, any and all Damages that Buyer Indemnified Parties incur as a result of, or with respect to (and whether or not in connection with any third-party claim):

(a) the inaccuracy or breach of any representation or warranty contained in or made pursuant to this Agreement;

(b) the non-compliance with or failure to perform any agreement or covenant of such Owner or Sellers contained in or made pursuant to this Agreement;

(c) any and all of Sellers', Owners' and the Companies' Transaction Expenses to the extent not paid concurrent with the Closing and any Transaction Bonuses to the extent not paid concurrent with the Closing;

(d) any and all of Sellers', Owners' and the Companies' Indebtedness to the extent not paid concurrent with the Closing;

(e) any Indemnified Taxes;

(f) any fraud, willful misconduct or criminal acts of such Owner, or, with respect to any period prior to the Closing Date, any Company;

(g) the matters specifically described on <u>Schedule 8.2(g)</u>; and

(h) any claim asserted by a third party relating to or resulting from any of the foregoing items (a) through (g).

8.3 <u>Indemnification by Buyer</u>. Subject to the other provisions of this <u>Article VIII</u>, Buyer agrees to indemnify, defend and hold harmless Seller Indemnified Parties from and against and shall pay to Seller Indemnified Parties the amount of, or reimburse Seller Indemnified Parties for, any and all Damages that Seller Indemnified Parties incur as a result of, or with respect to (and whether or not in connection with any third-party claim):

(a) the inaccuracy or breach of any representation or warranty contained in or made pursuant to this Agreement;

(b) the non-compliance with or failure to perform any agreement or covenant of Buyer contained in or made pursuant to this Agreement; and

(c) any claim asserted by a third party relating to or resulting from any of the foregoing items (a) through (b).

8.4 <u>Limitations</u>. Notwithstanding any other provision of this Agreement, the rights of the Parties to be indemnified and held harmless under this Agreement shall be limited as follows:

43

(a) Any claim for indemnification must be asserted before the expiration of the applicable Survival Period set forth in <u>Section 8.1</u>; provided, however, in the event written notice of any specifically identified claim for indemnification shall have been given to the Indemnifying Party within the applicable Survival Period, the provisions that are the subject of the indemnification claim shall survive with respect to such claims until such time as such claim is finally resolved.

(b) Other than with respect to claims relating to breaches of any Fundamental Representations or Special Representations:

(i) no Party shall have any Liability for Damages pursuant to <u>Section 8.2(a)</u> or <u>Section 8.3(a)</u>, as applicable, until the total of all Damages incurred by the Buyer Indemnified Parties or the Seller Indemnified Parties, as applicable, exceeds \$100,000, in which event the applicable Indemnifying Party shall only be liable for the Damages in excess of such threshold amount;

(ii) the Liability of each Owner to Buyer Indemnified Parties for Damages pursuant to <u>Section 8.2(a)</u> shall not exceed such Owner's Percentage Interest of 20% of the amount of the Purchase Price actually paid to Sellers; and

(iii) the Liability of Buyer to Seller Indemnified Parties for Damages pursuant to <u>Section 8.3(a)</u> shall not exceed an amount equal to 20% of the Purchase Price.

(c) With respect to claims relating to breaches of any Fundamental Representations or Special Representations, or based on fraud:

(i) the Liability of each Owner to Buyer Indemnified Parties for Damages pursuant to <u>Section 8.2(a)</u> shall not exceed such Owner's Percentage Interest of the amount of the Purchase Price actually paid to Sellers; and

(ii) the Liability of Buyer to Seller Indemnified Parties for Damages pursuant to <u>Section 8.3(a)</u> shall not exceed the Purchase Price.

(d) Subject to the other limitations described in this <u>Section 8.4</u>, and except for breaches by an individual Owner of any covenant that, by its nature, can be breached on an individual basis (including covenants described in <u>Section 6.3</u>), Buyer may only recover from an Owner such Owner's Percentage Interest of the collective Liability of Owners for Damages under this <u>Article VIII</u>. Buyer may bring a separate action against any one or more Owners with respect to any such Liability.

(e) In calculating amounts payable to an Indemnified Party under this Agreement, the amount of any Damages shall be determined without duplication of any other Damages for which an indemnification claim has been made under any other representation, warranty, covenant or agreement.

67350961.11

(f) In calculating amounts payable to any Indemnified Party under this Agreement, the amount of any Damages shall be determined net of any amounts actually recovered by any Indemnified Party under insurance policies (provided that, any deductible or co-payment or increase in premiums shall constitute additional Damages payable to the Indemnified Party) or under any third-party Contracts ("<u>Collateral Sources</u>") regarding such Damages, provided, that, nothing in this <u>Section 8.4(f)</u> shall (i) require an Indemnified Party to proceed or seek action or recovery from any Collateral Sources as a requirement hereunder or as a condition to seeking or recovering indemnification from any Indemnifying Party hereunder, or (ii) be construed or interpreted as a guaranty of any level or amount of insurance recovery with respect to any Damages hereunder or as a requirement to maintain any insurance or to make any claim for insurance as a condition to any indemnification hereunder.

Law.

(g) Any Indemnified Parties shall make reasonable efforts to mitigate any Damages in accordance with applicable

(h) No Indemnified Party will be entitled to indemnification pursuant to Section 8.2 for any (i) punitive Damages (except to the extent paid to third parties in connection with a third party claim or paid or awarded to any Governmental Authority) or (ii) Damages of a type that would not be reasonably foreseeable by the Parties as of the Effective Time or the Closing Date to result from a breach of any representation, warranty, covenant or other obligation or agreement of any Party hereunder.

(i) The Buyer Indemnified Parties will not be entitled to indemnification pursuant to Section 8.2 for Damages to the extent that any Buyer Indemnified Party has been previously compensated therefor pursuant to Section 2.3.

8.5 Indemnification Procedures.

(a) Whenever any indemnification claim shall arise in favor of a Person entitled to indemnification under this Article <u>VIII</u> (the "<u>Indemnified Party</u>"), including the assertion of any claim or liability against such Indemnified Party by a third party in writing that would give rise to a claim under this Article VIII, the Indemnified Party shall notify the Person giving the indemnity ("Indemnifying Party") in writing as soon as reasonably practicable but at least within fifteen (15) days of (i) such Indemnified Party receiving actual knowledge of the facts constituting the basis for such indemnification claim, or, (ii) in the case of a thirdparty claim, receipt of a written third-party assertion of a claim or liability. Failure to send such written notice shall not release the Indemnifying Party from liability hereunder, unless such failure materially prejudices the Indemnifying Party's defense of the claims that are the subject of the written notice.

(b) The Indemnifying Party shall have the option to assume the defense of any third-party claim and control the defense, settlement and prosecution of any litigation. Each Indemnified Party shall fully and reasonably cooperate with the Indemnifying Party in any such litigation defense, settlement or prosecution, and the Indemnifying Party shall reimburse each Indemnified Party for the actual out-of-pocket expenses incurred by the Indemnified Party as a result of such cooperation. The Indemnified Parties shall have the right to approve defense

67350961.11

counsel selected by the Indemnifying Party, which approval shall not be unreasonably withheld or delayed. The Indemnified Party shall be entitled to participate in the defense of such Action or claim, and employ separate counsel of its choice for such purpose. After notice from the Indemnifying Party to the Indemnified Party of its election to assume the defense of such Action or claim, the Indemnifying Party will not, as long as it diligently conducts such defense, be liable to the Indemnified Party under this Article VIII for any fees of other counsel or any other expenses with respect to the defense of such Action or Proceeding, in each case subsequently incurred by the Indemnified Party in connection with the defense of such Action or Proceeding, provided, that, if in the reasonable opinion of counsel to the Indemnified Party, (i) there are legal defenses available to an Indemnified Party that are different from or additional to those available to the Indemnifying Party; or (ii) there exists a conflict of interest between the Indemnifying Party and the Indemnified Party that cannot be waived, the Indemnifying Party shall be liable for the reasonable fees and expenses of counsel to the Indemnified Party in each jurisdiction for which the Indemnified Party determines counsel is required. Anything in this Section 8.5 notwithstanding, the Indemnifying Party shall not, without the written consent of the Indemnified Party, which consent shall not be unreasonably withheld or delayed, settle or compromise any claim or consent to the entry of any judgment. All Parties agree to cooperate fully as necessary in the defense of such matters, including making available records, information, personnel and testimony, and attending such conferences, discovery proceedings, hearings, trials or appeals relating to such third party claim and furnishing, without expense (other than reimbursement of actual out-of-pocket expenses) to the defending party, employees of the non-defending party as may be reasonably necessary for the preparation of the defense of such third-party claim.

8.6 <u>Disregarding Materiality Exceptions</u>. For purposes of determining whether a breach of any such representation or warranty has occurred, the terms "material," "materiality," and other qualifiers, modifiers or limitations shall be disregarded, provided, that the foregoing clause shall not apply to such determinations relating to breaches of Fundamental Representations or Special Representations.

8.7 <u>Exclusive Remedy</u>. The Parties acknowledge and agree that their sole and exclusive remedy with respect to any and all claims for any breach of any representation, warranty, covenant, agreement or other obligation set forth in this Agreement or otherwise relating to the subject matter of this Agreement, shall be pursuant to the indemnification provisions set forth in this <u>Article VIII</u>. Nothing in this <u>Section 8.6</u> shall limit the Parties' right to (a) seek and obtain any equitable or injunctive relief to which any Party may be entitled, (b) utilize the dispute resolution provisions described in <u>Section 2.3</u> or (c) seek any remedy on account of any Party's fraudulent, criminal or willful misconduct.

67350961.11

ARTICLE IX - NOTICES

All notices and other communications hereunder shall be in writing and shall be given to the Parties via email, hand delivery, or nationally recognized and reputable overnight delivery service, addressed to the Parties as follows:

Buyer: i3-Bearcat, LLC

c/o i3 Verticals, LLC 40 Burton Hills Boulevard Suite No. 415 Nashville, Tennessee 37215 Attention: General Counsel

with copies to: Frost Brown Todd LLC 150 Third Avenue South Suite No. 1900 Nashville, Tennessee 37201 Attention: Howard W. Herndon Facsimile: 615-251-5551 Email: HHerndon@FBTlaw.com

Sellers or Owners: David Graves 1660 Loop 301 Sulphur Springs, Texas 75482

with copies to: Polsinelli PC 2950 N. Harwood Suite 2100 Dallas, Texas 75201 Attention: Jonathan K. Henderson Email: jhenderson@Polsenelli.com

Each such notice and other communication shall be deemed, for all purposes of this Agreement, to have been given and received (a) if given by email, when the email is transmitted to the Party's email address specified above and confirmation of complete receipt is received by the transmitting Party during normal business hours on any Business Day or on the next Business Day if not confirmed during normal business hours; (c) if by hand, when delivered; or (c) if given by nationally recognized and reputable overnight delivery service, the Business Day on which the records of such delivery service show that such notice was delivered to the Party. Any Party from time to time may change its address or facsimile number for the purpose of receipt of notices to that Party by giving a similar notice specifying a new address or facsimile number to the other notice Parties listed above in accordance with the provisions of this <u>Article IX</u>.

67350961.11

ARTICLE X - PROVISIONS CONCERNING THE REPRESENTATIVE

10.1 <u>Appointment of Representative</u>. Each Seller and each Owner hereby irrevocably appoints David Graves as the agent, proxy and attorney-in-fact for such Sellers and such Owner for all purposes of this Agreement and all transactions, documents and other agreements contemplated herein, with full and exclusive power and authority to act on Sellers' and such Owner's behalf as their representative (the "<u>Representative</u>"). The appointment of the Representative hereunder is coupled with an interest, and shall not be affected by the death, incapacity, insolvency, bankruptcy, illness or other inability to act of any Seller or other Owner. The Representative hereby accepts such appointment. Without limiting the generality of the foregoing, the Representative is hereby authorized, on behalf of Sellers and Owners, to:

(a) in connection with the Closing, execute, deliver and receive all documents, instruments, certificates, statements and agreements on behalf of and in the name of each of the Sellers and each of the Owners necessary to effectuate the Closing and consummate the Contemplated Transactions, including the distribution of the Purchase Price;

(b) receive and give all notices and service of process, make all filings, enter into all contractual obligations, make all decisions, bring, prosecute, defend, settle, compromise or otherwise resolve all claims, disputes and actions, authorize payments in respect of any such claims, disputes or actions, and take all other actions directly or indirectly arising out of or relating to this Agreement and all transactions, documents and other agreements contemplated herein; and

(c) take all other actions permitted or required to be taken by or on behalf of Sellers and Owners under this Agreement and exercise any and all rights that Sellers, Owners or the Representative are permitted or required to do or exercise under this Agreement.

10.2 <u>No Liability</u>. The Representative shall not be held liable by any Seller or any Owner for actions or omissions in exercising or failing to exercise all or any of the power and authority of the Representative pursuant to this Agreement, except in the case of the Representative's gross negligence, bad faith or willful misconduct. The Representative shall be entitled to rely on the advice of counsel, accountants or other independent experts that it reasonably determines to be experienced in the matter at issue, and will not be liable to any Seller or any Owner for any action taken or omitted to be taken in good faith based on such advice. Owners will indemnify the Representative from any Damages arising out of its serving as the Representative hereunder, except for Damages arising out of or caused by the Representative's gross negligence, bad faith or willful misconduct.

10.3 <u>Reliance on Representative</u>. Buyer may rely on the appointment and authority of the Representative granted pursuant to this <u>Article X</u>. In so doing, Buyer may rely on any and all actions taken by and decisions of the Representative under this Agreement notwithstanding any dispute or disagreement among any of the Sellers or Owners with respect to any such action or decision, without any Liability to, or obligation to inquire of, any of the Sellers or Owners or any other Person. Any decision, act, consent or instruction of the Representative shall constitute a decision of all Sellers and Owners and shall be final and binding upon each Seller and Owner. At

67350961.11

any time, with or without cause, by a written instrument that is signed in writing by Sellers and Owners and delivered to Buyer, Sellers and Owners may temporarily substitute and designate an alternative Representative. At any time, with or without cause, by a written instrument that is signed in writing by Sellers and Owners and delivered to Buyer, Sellers and Owners may remove and designate a successor Representative. If the Representative shall at any time be removed, resign or otherwise cease to function in its capacity as such for any reason whatsoever, and no successor is appointed by Sellers and Owners, then all references in this Agreement to the "Representative" shall be treated as references to all Sellers and Owners acting jointly in concert.

10.4 <u>Miscellaneous</u>. All payments to Owners on behalf of Sellers under this Agreement shall be made in proportion to the Percentage Interest of each Owner, and each Owner agrees to and acknowledges its respective Percentage Interest as the sole mechanism for determining its respective right, title and interest in and to the payment to which it is entitled in respect of its Equity Interests, as applicable (notwithstanding anything in any organizational document of any Company or any other Contracts related to the Equity Interests). The Representative shall not be entitled to any fee, commission or other compensation for the performance of its service hereunder, but shall be authorized to seek reimbursement from Sellers and Owners for any costs, expenses or other fees that may be paid or borne by the Representative in acting as the Representative under this Agreement.

ARTICLE XI - MISCELLANEOUS

11.1 <u>Fees and Expenses</u>. Except as otherwise expressly provided in this Agreement, each Party will pay its own costs and expenses incurred in connection with the negotiation, preparation and performance of this Agreement and the other Transaction Documents, including the fees and expenses of its counsel, accountants, brokers and financial advisors whether or not such transactions are consummated.

11.2 <u>Public Announcement</u>. No Party shall, prior to or after the Closing, without the approval of the other Parties, issue any press release or other public announcement concerning this Agreement or the Contemplated Transactions. Notwithstanding the foregoing, i3V may issue a press release or other communication on or following the Closing Date to announce the closing of the Contemplated Transaction, to provide a summary narrative about the Companies, the Business and the material terms of the Contemplated Transaction, or in connection with i3V's compliance with any applicable Law or pursuant to any rules or regulations of the Securities and Exchange Commission or any securities exchange of which the securities of i3V are listed or traded.

11.3 <u>Entire Agreement</u>. This Agreement (together with the Schedules and the other Transaction Documents) contains the entire agreement and understanding of the Parties with respect to the subject matter hereof and supersedes all prior oral discussions and written agreements with respect thereto (including any term sheet or similar agreement or document relating to the Contemplated Transactions). There are no restrictions, agreements, promises, warranties, covenants or undertakings other than those expressly set forth herein and in the other Transaction Documents.

67350961.11

11.4 <u>Amendment and Waiver</u>. This Agreement may be modified, supplemented or amended only by a written instrument duly executed by each of the Parties. Any term or condition of this Agreement may be waived at any time by the Party entitled to the benefit thereof. Any such waiver must be in writing and must be duly executed by such Party. All rights and remedies of the Parties to this Agreement are cumulative and not alternative. No failure or delay by any Party in exercising any right, power or privilege under this Agreement or the other Transaction Documents will operate as a waiver of such right, power or privilege, and no single or partial exercise of any such right, power or privilege will preclude any other or further exercise of such right, power or privilege or the exercise of any other right, power or privilege. A waiver on one occasion shall not be deemed to be a waiver of the same or any other breach, provision or requirement on any other occasion.

11.5 <u>Counterparts; Electronic Signatures</u>. This Agreement may be executed simultaneously in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Signatures on this Agreement by facisimile or other electronic imaging technology shall be deemed to be original signatures for all purposes.

11.6 Dispute Resolution.

(a) <u>Agreed Process</u>. Except for claims for injunctive or equitable relief or suit to compel specific performance under this Agreement as described in <u>Section 11.12</u> (including suits to compel compliance with the dispute resolution process set forth in this <u>Section 11.6</u>), any dispute arising out of or in connection with this Agreement or its validity shall be finally resolved and settled by arbitration in accordance with this <u>Section 11.6</u>. With respect to any reference to a "party" or "parties" under this <u>Section 11.6</u>, such references shall refer to Buyer, on the one hand, and Representative, on behalf of Sellers and Owners, on the other hand, as applicable. For purposes of clarity, any disputes between the parties under <u>Article VIII</u> (Indemnification) shall be referred to arbitration as described in this <u>Section 11.6</u>.

(b) <u>Arbitration</u>. Any arbitration to be conducted hereunder shall be brought and conducted in the city of Dallas, Texas and shall be governed by the Laws described in <u>Section 11.7</u>.

(c) Rules Governing the Conduct of the Arbitration.

(i) <u>Governing Rules</u>. Any dispute that is referred to arbitration under this <u>Section 11.6</u> shall be finally resolved by binding arbitration administered by the American Arbitration Association ("<u>AAA</u>") under its Commercial Arbitration Rules.

(ii) <u>Arbitration Panel</u>. The arbitration shall be presided over by a panel of three (3) arbitrators.

(iii) <u>Arbitrator Selection Procedure</u>. Within twenty (20) days after the claimant party provides the respondent party with the first notice of a demand for arbitration (the "<u>Demand Notice</u>"), each of Representative and Buyer shall select one (1) impartial person to

50

serve as an arbitrator. The two (2) arbitrators so appointed shall then select the presiding arbitrator within thirty (30) days after completion of the appointment of the party selections. If the two arbitrators are unable to agree on the presiding arbitrator within such thirty (30) day period, the AAA shall select the presiding arbitrator. The presiding arbitrator shall have at least fifteen (15) years of commercial litigation and dispute resolution experience either as an attorney, arbitrator, or judge.

(iv) <u>Pre-hearing Information Exchange</u>. Consistent with the expedited nature of arbitration, pre-hearing information exchange shall be limited to the reasonable production of relevant non-privileged documents explicitly referred to by a party for the purpose of supporting relevant facts presented in its case, carried out expeditiously. The parties shall be required to produce, if requested by another party, up to five (5) witnesses designated by the requesting party for depositions in locations convenient for the witnesses, unless the party opposing discovery persuades the arbitral tribunal that such discovery is unnecessary or would be unduly expensive in the context of the dispute.

(v) <u>Presentation of Witnesses</u>. At the hearing, either party may elect to present the direct testimony of any of its witnesses through live or video testimony or through written statements.

(vi) <u>Rulings</u>. Any interim or final arbitration award shall be accompanied by a reasoned opinion. The arbitrators' authority shall be limited to enforcing the terms of, and their decision shall be limited to granting only the relief contemplated under, this Agreement. Except as may be required by law, neither a party nor an arbitrator may disclose the existence, content, or results of any arbitration hereunder without the prior written consent of both parties.

(d) <u>Deadlines for Resolving Particular Disputes</u>. Regardless of the amount in controversy, all other disputes raised in the Demand Notice shall be finally decided within nine (9) months following the Demand Notice, unless such time limit is extended by the arbitrators.

(e) <u>Binding Arbitration</u>. Any arbitration award shall bind the parties, and judgment on the final award may be entered in any court of competent jurisdiction. The decision of the arbitrators shall be carried out voluntarily and without delay.

(f) <u>Costs and Fees</u>. Each Party shall bear its own costs and expenses and an equal share of the arbitrators' and administrative fees of arbitration.

(g) <u>Confidentiality</u>. For purposes of clarity, the existence of any dispute under this Agreement, the fact that an arbitration is being conducted and all information concerning the arbitration described in this <u>Section 11.6</u> or shared during such arbitration shall be subject to the confidentiality provisions set forth in <u>Section 7.3</u>.

11.7 <u>Governing Law; Construction</u>. This Agreement shall be governed by and interpreted, construed and enforced in accordance with the Laws of the State of Delaware applicable to a contract executed and performed in such state, excluding any conflicts of law, rule

67350961.11

or principle that would refer the governance, interpretation, construction or enforcement of this Agreement to the Laws of another jurisdiction, and such application of Delaware law shall not be vitiated by any allegations of fraud. In as much as this Agreement is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, the Parties agree that no provisions of this Agreement or any related document shall be construed for or against or interpreted to the advantage or disadvantage of any Party hereto by any court or other Governmental Authority by reason of any Party's having or being deemed to have structured or drafted such provision, each Party having participated equally in the structuring and drafting hereof.

11.8 <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY AND ALL RIGHTS TO TRIAL BY JURY, FROM WHATEVER SOURCE ARISING, IN CONNECTION WITH ANY LITIGATION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

11.9 <u>Binding Effect; No Assignment; No Third-Party Beneficiaries</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns, including successors by merger or otherwise. Neither this Agreement nor any right hereunder or part hereof may be assigned by any Party hereto without the prior written consent of the other Parties, except that Buyer may assign this agreement and its rights hereunder to an affiliate of Buyer or to a person or entity that acquires or otherwise succeeds to the Business (whether by sale of assets, equity, merger, or otherwise). The terms and provisions of this Agreement are intended solely for the benefit of Sellers, Owners, the Companies, Buyer, i3 LLC, i3V and their respective successors or permitted assigns, and it is not the intention of the Parties to confer third-party beneficiary rights upon any other Person.

11.10 <u>Severability</u>; <u>Invalid Provisions</u>. It is the intention of the Parties that the provisions of this Agreement shall be enforced to the fullest extent permissible under the Laws and public policies of each state and jurisdiction in which such enforcement is sought, and that the unenforceability (or the modification to conform with such Laws or public policies) of any provision hereof shall not render unenforceable or impair the remainder of this Agreement. Accordingly, if any provision of this Agreement is held to be illegal, invalid or unenforceable under any present or future Law, (a) such provisions will be fully severable; (b) this Agreement will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof; (c) the remaining provisions of this Agreement will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom; and (d) in lieu of such illegal, invalid or unenforceable provision as a part of this Agreement a legal, valid and enforceable provision as similar in terms and effect to such illegal, invalid or unenforceable provision as may be possible (or, in the alternative, should any provision contained in this Agreement be reformed or rewritten by any Governmental Authority, such provision as so reformed shall be fully binding on the Parties as if originally a part hereof).

67350961.11

11.11 Interpretation. In this Agreement, unless the context otherwise requires:

(a) subject to the provisions of <u>Section 11.9</u>, references to any Party to this Agreement shall include references to its respective successors and permitted assigns;

(b) the terms "<u>hereof</u>," "<u>herein</u>," "<u>hereby</u>," and derivative or similar words will refer to this entire Agreement;

(c) the gender of all words herein shall include the masculine, feminine and neuter, and the case of all words herein shall include the singular and plural;

(d) references to any document (including this Agreement) are references to that document as amended, consolidated, supplemented, novated or replaced by the Parties from time to time;

(e) the descriptive headings and numbers of the Articles, Sections and subsections of this Agreement are inserted for convenience only and do not constitute a part of this Agreement;

(f) the word "including" shall mean "including, without limitation"; and

(g) all schedules and exhibits referred to in or attached to this Agreement are integral parts of this Agreement as if fully set forth herein.

11.12 <u>Specific Performance; Injunctive Relief</u>. Each Party shall have the right, in addition to any other rights and remedies existing in its favor, to enforce its rights and the obligations of the other Parties pursuant to this Agreement and the Transaction Documents not only by an action or actions for Damages, but also by an action or actions for specific performance, injunctive relief and/or other equitable relief, without posting any bond or other undertaking. The Parties acknowledge and agree that any breach or threatened breach of any post-Closing covenant by any Party will likely result in some irreparable injury.

11.13 <u>Collateral Security</u>. Buyer may, without the written consent of the other Parties, assign its rights under this Agreement and the other Transaction Documents for collateral security purposes to any lender providing financing to Buyer or any of their affiliates and any such lender may exercise all of the rights and remedies of Buyer hereunder and thereunder.

11.14 Representation by Counsel.

(a) Each Party represents and agrees with each other Party that it has been represented by or had the opportunity to be represented by, independent counsel of its own choosing, and that it has had the full right and opportunity to consult with its respective attorney(s), that to the extent, if any, that it desired, it availed itself of this right and opportunity, that it or its authorized officers (as the case may be) have carefully read and fully understand this Agreement in its entirety and have had it fully explained to them by such Party's respective counsel, that each is fully aware of the contents thereof and its meaning, intent and legal effect,

53

and that it or its authorized officer (as the case may be) is competent to execute this Agreement and has executed this Agreement free from coercion, duress or undue influence.

(b) Each of the Parties hereby agrees, on its own behalf and on behalf of its representatives, that (a) the law firm of Polsinelli PC (the "Company's Counsel") may have served as counsel to each of the Representative, Sellers, Owners and any affiliate of Owners, on the one hand, and the Companies, on the other hand, in connection with the negotiation, preparation, execution, and delivery of this Agreement and the consummation of Contemplated Transactions; and (b) following consummation of Contemplated Transactions, the Company's Counsel (or any of its successors) may serve as counsel to the Representative, Sellers, Owners or any of their respective affiliates or any of their respective representatives, in connection with any Action, claim or obligation arising out of or relating to this Agreement or the Contemplated Transactions or any other matter, and each of the Parties hereby consents thereto and waives any conflict of interest arising therefrom, and each of such Parties shall cause any representative thereof to consent to waive any conflict of interest arising from such representation. Each of the Parties further agrees to permit (and shall take reasonable steps requested by any Party at such requesting party's expense so that) any privilege attaching as a result of the services provided by the Company's Counsel as counsel to the Representative, Sellers, Owners (including with respect to information related to the Companies), and the Companies in connection with the Contemplated Transactions to survive the Closing and to remain in effect, and such privilege shall continue to be controlled solely by the Representative following the Closing. In addition, all of the client files and records in the possession of the Company's Counsel related to the Contemplated Transactions to the extent constituting attorney work-product or any other privileged or confidential documents or information shall continue to be property of (and be controlled by) by the Representative, Sellers and Owners, and the Companies shall not retain any copies of such records or have any access to them.

11.15 <u>Parent Guaranty</u>. i3 LLC and Buyer will be jointly and severally liable for Buyer's payment to Sellers of all payments of the Purchase Price and all other payments owing by Buyer pursuant to <u>Sections 2.2</u>, <u>2.3</u> and <u>Article VIII</u>. Notwithstanding anything in this Agreement to the contrary, i3 LLC shall have, and shall be entitled to assert or exercise, all of the defenses, remedies and other rights available to Buyer under this Agreement or any applicable Law and shall receive copies of all notices required to be delivered to Buyer under the terms of this Agreement at the time so required hereunder, simultaneously with delivery to Buyer.

[Signature pages follow.]

54

IN WITNESS WHEREOF, the Parties have caused this Membership Interest Purchase Agreement to be executed as of the date first above written.

BUYER:

i3-BEARCAT, LLC

By: <u>/s/ Rick Stanford</u> Name: <u>Rick Stanford</u> Title: <u>President</u>

<u>i3 LLC</u>:

Solely with respect to Section 11.15:

i3 VERTICALS, LLC

By: <u>/s/ Rick Stanford</u> Name: <u>Rick Stanford</u> Title: <u>President</u>

SELLERS:

NTD HOLDINGS, INC.

By: <u>/s/ David V. Graves</u> Name: <u>David V. Graves</u> Title: <u>President</u>

GH HOLDCO, INC.

By: <u>/s/ David V. Graves</u> Name: <u>David V. Graves</u> Title: <u>President</u>

OWNERS:

DAVID GRAVES

/s/ David Graves

TORY HUMPHRIES

/s/ Tory Humphries

SCHEDULE 1

DEFINITIONS

As used in this Agreement, the following terms shall have the meanings set forth in this <u>Schedule 1</u> unless the context clearly otherwise requires.

"<u>AAA</u>" has the meaning set forth in <u>Section 11.6(c)(i)</u>.

"Acquired Business" means the Companies and the Business as operated by Buyer following the Closing.

"<u>Action</u>" means any claim, action, cause of action, demand, lawsuit, arbitration, inquiry, audit, notice of violation, proceeding, litigation, citation, summons, subpoena or investigation of any nature, civil, criminal, administrative, regulatory or otherwise, whether at law or in equity.

"<u>Adjustment Calculation</u>" means an amount, which may be positive or negative, equal to the aggregate positive or negative differences between (a) the Net Working Capital Calculation compared to the Estimated Net Working Capital; (b) the Closing Indebtedness compared to Estimated Indebtedness; and (c) the Closing Transaction Expenses compared to the Estimated Transaction Expenses. For purposes of clarity, (i) an excess of a positive Net Working Capital Calculation over the Estimated Net Working Capital would generate a positive adjustment; (ii) an excess of the Closing Transaction Expenses over the Estimated Indebtedness would generate a negative adjustment; and (iii) an excess of the Closing Transaction Expenses over the Estimated Transaction Expenses would generate a negative adjustment.

"<u>Agreement</u>" has the meaning set forth in the Preamble.

"<u>Benefit Plans</u>" means "employee benefit plans," as defined in Section 3(3) of ERISA, and all other stock purchase, stock option, equity-based, retention bonus, bonus, incentive compensation, deferred compensation, profit sharing, severance, change in control, supplemental unemployment, layoff, salary continuation, retirement, pension, health, life insurance, disability, group insurance, vacation, holiday, sick leave, fringe benefit, welfare and other employee benefit plans or employment (including severance and change in control) agreement, program, policy or other arrangement (whether formal or informal, oral or written, qualified or non-qualified, and whether or not subject to ERISA), including any funding mechanism therefore, under which any employee or former employee of any Company has any present or future right to benefits or under which any Company has any present or future Liability.

"<u>Books and Records</u>" means any Company's existing accounting, business, marketing, personnel, and other files, documents, instruments, papers, books and records, including, financial statements, budgets, ledgers, journals, deeds, titles, policies, manuals, minute books, Contracts, franchises, permits, supplier lists, reports, computer files and data, retrieval programs and operating data or plans; excluding however, minute books, stock certificates and books and stock transfer ledgers.

"Business" has the meaning set forth in the Recitals.

"<u>Business Day</u>" means a day other than Saturday, Sunday, or any day on which commercial banks located in the State of New York are authorized or obligated to close under the Laws of such state or the United States.

"Buyer" has the meaning set forth in the Preamble.

"<u>Buyer Indemnified Parties</u>" means Buyer and any of its affiliates (including, with respect to any post-Closing period, the Companies) and the equity holders, members, shareholders, directors, managers, officers, employees, agents, representatives, successors and assigns and affiliates of any of the foregoing.

"<u>Card Brand</u>" means any of VISA, U.S.A., Mastercard International Inc., American Express, DFS Services, LLC (Discover) and all other applicable credit card issuers or networks.

"Class V Excess Allocation" has the meaning set forth in Section 6.2(j).

"Class VI Excess Allocation" has the meaning set forth in Section 6.2(j).

"Closing" means the consummation of the Contemplated Transactions as provided in Section 2.4.

"<u>Closing Payment</u>" has the meaning set forth in <u>Section 2.6(a)</u>.

"Closing Cash Purchase Price" has the meaning set forth in Section 2.2(a).

"<u>Closing Date</u>" has the meaning set forth in <u>Section 2.4</u>.

"<u>Closing Memorandum</u>" has the meaning set forth in <u>Section 2.5(e)</u>.

"<u>COBRA</u>" means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, Section 4980B of the Code, Title I, Part 6 of ERISA and the Public Health Service Act, together with all regulations promulgated thereunder.

"Code" means the Internal Revenue Code of 1986, as amended, and the rules and regulations promulgated thereunder.

"<u>Company Intellectual Property</u>" means all Intellectual Property that is used, is held for use or is contemplated to be used by any Company.

"<u>Company Intellectual Property Registrations</u>" means all of the registrations for Intellectual Property owned by or under obligation of assignment to or applications filed in the name of any Company or otherwise pertaining to any Company Intellectual Property.

"<u>Company Product</u>" means any software, application or product that any Company develops, manufactures, distributes, sells, or provides access to, including the Platform.

"<u>Contemplated Transactions</u>" means the transactions contemplated by this Agreement and the other Transaction Documents.

"<u>Contracts</u>" means all commitments, contracts, leases, subleases, licenses, sublicenses, subscriptions, agreements for rebates (whether or not free-standing or part of any of the foregoing) and other agreements of any kind to which any Company is a party or by which any Company is bound, in each case regardless of whether or not a written document is in force with respect thereto.

"<u>Copyleft License</u>" means any software license that requires, as a condition of use, modification or distribution of software, that such software be subject to such license (a) be made available or distributed in a form other than binary (e.g., source code form); (b) be licensed for the purpose of preparing derivative works; (c) be licensed under terms that allow Company Products or portions thereof or interfaces therefor to be reverse engineered, reverse assembled or disassembled (other than by operation of Law); or (d) be redistributable at no license fee. Copyleft Licenses include the GNU General Public License, the GNU Lesser General Public License, the Mozilla Public License, the Common Development and Distribution License, the Eclipse Public License and all Creative Commons "sharealike" licenses.

"<u>Copyleft Materials</u>" means any software or other Intellectual Property subject to a Copyleft License.

"Cryptocurrencies" has the meaning set forth in Section 4.26.

"<u>Current Assets</u>" means cash and cash equivalents, accounts receivable, inventory and prepaid expenses, but <u>excluding</u> (a) the portion of any prepaid expense of which Buyer will not receive the benefit following the Closing, (b) deferred Tax assets and (c) receivables from any of the Company's affiliates, directors, employees, officers or members and any of their respective affiliates, in each case, determined in accordance with GAAP (except with respect to the Estimated Closing Statement as contemplated by <u>Section 2.3(a)</u>).

"<u>Current Liabilities</u>" means accounts payable, accrued Taxes, accrued expenses, deferred revenues (including long-term deferred revenue, even if inconsistent with GAAP), unearned revenues, customer deposits or any other similar liabilities for payment received in advance for services which have not yet been performed or goods which have not yet been delivered, earned-but-unused paid time off and sick leave, and accrued interest, but <u>excluding</u> Transaction Expenses and any Indebtedness of the Company, in each case determined in accordance with GAAP (except with respect to the Estimated Closing Statement as contemplated by <u>Section 2.3(a)</u>).

"<u>Customer</u>" means a Person, including any "merchants" (as such term is generally used in the payment processing industry), county government, city court system, or other entity, (i) from which any Company directly or indirectly receives compensation as a result of such Company selling, reselling, licensing or servicing products, equipment or software including any Company Products, (ii) for whom any Company provides or enables payment processing or other products and services relating to the Business or (iii) with respect to whom any Company

receives residuals, commissions or fees, including processing or enabling ACH, credit or debit card payments, in all cases including those Persons set forth on <u>Schedule 1.1(a)</u>.

"<u>Customer Agreement</u>" means, for any Customer, any agreement between such Customer on the one hand and one or more of a Company, a Processor, a merchant bank and/or a Referral Source on the other.

"<u>Damages</u>" means losses, Liabilities, damages, costs (including court costs, costs of appeal and, costs with respect to enforcement of an indemnity claim and costs of pursuing any insurance providers), Taxes and expenses (including reasonable attorney's fees), whether accruing before or after the expiration of any applicable Survival Period.

"Data Breach" has the meaning set forth in Section 4.25.

"Demand Notice" has the meaning set forth in Section 11.6.

"<u>Disclosure Schedules</u>" means the Disclosure Schedules delivered by Sellers to Buyer concurrently with the execution and delivery of this Agreement.

"<u>Disregarded Entity</u>" means an entity, other than one which is a QSub, which is a limited liability company which is disregarded as an entity separate from the Companies for U.S. federal and state income tax purposes for which no election under Treasury Regulation Section 301.7701-3(c)(1)(i) for such entity is required to be treated in any other manner (including the filing of an Internal Revenue Service Form 8832).

"Domain Names" means internet domain names and numbers.

"Effective Time" has the meaning set forth in the Preamble.

"<u>Employee Non-Competition, Non-Solicitation, Confidentiality and Inventions Agreement</u>" means Buyer's or its affiliate's standard employee restrictive covenant and inventions assignment agreement.

"End Users" are Persons who are the end users of any Company Product.

"<u>Environmental Claim</u>" means any investigation or written claim, action, cause of action, or notice by any person alleging potential Liability (including potential Liability for investigatory costs, cleanup costs, governmental response costs, natural resources damages, property damages, personal injuries, or penalties) arising out of, based on or resulting from: (a) the presence, or release or threat of release into the environment, of any Materials of Environmental Concern at any location owned or operated by any Company; or (b) circumstances forming the basis of any violation or alleged violation of any Environmental Law applicable to any Company or the Business.

"<u>Environmental Laws</u>" means as of the Closing Date, all applicable Laws relating to pollution or protection of human health (as relating to the environment or the workplace) and the environment (including ambient air, surface water, ground water, land surface or sub-surface

strata), including Laws relating to emissions, discharges, releases or threatened releases of Materials of Environmental Concern, or otherwise relating to the use, treatment, storage, disposal, transport or handling of Materials of Environmental Concern, including, but not limited to Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq., Resource Conservation and Recovery Act, 42 U.S.C. § 6901 et seq., Toxic Substances Control Act, 15 U.S.C. § 2601 et seq., Occupational Safety and Health Act, 29 U.S.C. § 651 et seq., the Clean Air Act, 42 U.S.C. § 7401 et seq., the Clean Water Act, 33 U.S.C. § 1251 et seq., each as may have been amended or supplemented, and any applicable environmental transfer statutes or Laws.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"<u>ERISA Affiliate</u>" means: (a) any related company or trade or business that is required to be aggregated with any Company under Code Sections 414(b), (c), (m) or (o); (b) any other company, entity or trade or business that has adopted or has ever participated in any Benefit Plan related to any Company; and (c) any predecessor or successor company or trade or business of any Company.

"Estimated Closing Statement" has the meaning set forth in Section 2.3(a).

"Estimated Indebtedness" has the meaning set forth in Section 2.3(a).

"Estimated Net Working Capital" has the meaning set forth in Section 2.3(a).

"<u>Estimated Transaction Expenses</u>" has the meaning set forth in <u>Section 2.3(a)</u>.

"Equity Interests" means 100% of the membership interests in the Companies.

"<u>Final Adjustment Calculation</u>" means an amount equal to (i) the Adjustment Calculation if Buyer does not receive an Objection Notice pursuant to and in accordance with <u>Section 2.3(c)</u> or (ii) otherwise, the aggregate positive or negative differences between (A) the Final Net Working Capital Calculation compared to the Estimated Net Working Capital; (B) the Final Indebtedness compared to the Estimated Indebtedness; and (C) the Final Transaction Expenses compared to the Estimated Transaction Expenses.

"Final Indebtedness" has the meaning set forth in <u>Section 2.3(d)</u>.

"Final Net Working Capital Calculation" has the meaning set forth in Section 2.3(d).

"<u>Final Transaction Expenses</u>" has the meaning set forth in <u>Section 2.3(d)</u>.

"Financial Statements" has the meaning set forth in Section 4.5.

"<u>Fundamental Representations</u>" means those representations and warranties set forth in <u>Section 3.1</u>, <u>Section 3.4</u>, <u>Section 4.2</u>, <u>Section 4.8</u>, <u>Section 4.12(a)</u>, <u>Section 5.1</u>, <u>Section 5.2</u> and <u>Section 5.4</u>.

"<u>Governmental Authority</u>" means the government of the United States and any government of a state or other political subdivision thereof, any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including any government authority, agency, department, board, commission or instrumentality of the United States (including receivers or other agents appointed by any of the foregoing), any state of the United States or any political subdivision thereof, any tribunal or arbitrator(s) of competent jurisdiction and any self-regulatory organization.

"Gross Up Payment" has the meaning set forth in Section 6.2(j).

"<u>Historical Accounting Principles</u>" means the accounting methods, policies, practices, procedures and assumptions used by the Companies in the preparation of the Companies' historical financial statements. Any material differences between such methods, policies, practices, procedures and assumptions and GAAP are described in <u>Schedule 4.5(b)</u>.

"<u>i3V</u>" means i3 Verticals, Inc., a Delaware corporation.

"<u>i3V Business</u>" means the provision of (i) payment processing and related products and services (including a payment gateway, point-of-sale equipment, etc.), and (ii) proprietary and third party software products that contain an integrated payment processing component.

"Income Tax Proceeding" has the meaning set forth in Section 6.2(e).

"Indebtedness" means, with respect to any Person, all Liabilities, contingent or otherwise, as obligor or otherwise, for or with respect to (a) all obligations for borrowed money, including as arising from corporate credit cards, which obligations shall include the principal, accreted value, accrued and unpaid interest, unpaid fees or expenses and other monetary obligations or other interest-bearing indebtedness, whether current or funded, secured or unsecured; (b) all obligations evidenced by a note, bond or debenture; (c) all obligations for deferred purchase price of any property or services; (d) all obligations created or arising under any conditional sale or other title retention agreement with respect to property acquired, including any "earnout" or similar payments or any noncompete payments; (e) all obligations secured by a purchase money mortgage or other lien to secure all or part of the purchase price of property subject to such mortgage or lien; (f) all obligations in respect of bankers' acceptances, letters of credit or similar credit transactions; (g) all obligations secured by Liens on property acquired, whether or not such obligations were assumed at the time of acquisition of such property; (h) any off balance sheet financial obligations in the nature of indebtedness, including synthetic leases and project financing; (i) the face value of any surety bonds, performance bonds or security deposits; (j) breakage or similar costs for interest rate hedges or early termination of any of the obligations of a type reflected above; (k) all obligations of any Company to any Seller, Owner, manager, member, officer or employee of any of the Companies or to any affiliate of the foregoing other than Liabilities to current employees of any of the Companies incurred in the ordinary course of business; (1) all obligations of a type referred to above which are directly or indirectly guaranteed by any of the Companies or which any of the Companies has agreed (contingently or otherwise) to purchase or otherwise acquire or in respect of which it has otherwise assured a credit against loss; or (m) all other Liabilities, that, in accordance with GAAP, should be classified upon the

balance sheet of any of the Companies as indebtedness. For purposes of this Agreement, Indebtedness includes the aggregate amount of any accrued interest, accreted value, breakage costs, prepayment premiums or penalties related thereto, unpaid fees or other costs or expenses associated with the prepayment or termination of any Indebtedness. Notwithstanding the foregoing, Indebtedness shall not include any amounts that are included or taken into account in the calculation of Net Working Capital.

"Indemnified Party" has the meaning set forth in Section 8.5(a).

"Indemnified Taxes" means, without duplication, any of the following Taxes (in each case, whether imposed, assessed, due or otherwise payable directly, as a successor or transferee, jointly and/or severally, pursuant to a contract or other agreement entered into (or assumed) by any of the Companies on or prior to the Closing Date, or for any other reason and whether disputed or not): (a) any Tax in respect of the Companies for any Pre-Closing Tax Period or portion of a Straddle Tax Period ending on or before the Closing Date to the extent not included in the Net Working Capital Calculation; (b) any Tax that any Company is liable for (including under Section 1.1502-6 of the Treasury Regulations or any similar provision of state, local, or foreign applicable Laws) as a result of being a member of (or leaving) an affiliated group on or before the Closing Date or any similar provisions of federal, state or local law imposing successor liability for Taxes; (c) any Tax resulting from a breach of a representation or warranty in Section 4.15 (Employee Benefit Plans), Section 4.16 (Employee Relations) or Section 4.17 (Taxes), or a breach of a covenant of Sellers in Section 6.2; (d) any and all taxes of any person imposed on a Company arising under the principles of transferee or successor liability or by contract, relating to an event or transaction occurring before the Closing Date, (e) the Companies' share of any Taxes arising from the payment of any Transaction Bonuses, and (f) any Taxes resulting from the inclusion of any material item of income in, or exclusion of any material item of deduction from, taxable income for any period (or any portion thereof) ending after the Closing Date as a result of any (i) installment sale or other transaction on or prior to the Closing Date, (ii) tax closing agreement pursuant to Section 7121 of the Code or any corresponding provision of state, local or foreign Tax Law, entered into on or prior to the Closing Date, (iii) accounting method change or agreement with any Governmental Authority, (iv) prepaid amount received on or prior to the Closing Date, (v) income from discharge of indebtedness realized on or prior to the Closing Date, pursuant to Section 108(i) of the Code or any corresponding provision of state, local or foreign Tax Law, in each case, to the extent not taken into account in the Net Working Capital Calculation, or (vi) election by the Companies to defer any taxes attributable to a deemed repatriation of foreign assets under applicable U.S. federal, state or local Tax Law.

"<u>Indemnifying Party</u>" has the meaning set forth in <u>Section 8.5(a)</u>.

"<u>Intellectual Property</u>" means all intellectual property and other similar proprietary rights in any jurisdiction worldwide, whether owned or held for use under license, whether registered or unregistered, including such rights in and to: (i) Trademarks; (ii) inventions, invention disclosures, discoveries and improvements, whether or not patentable; (iii) issued patents and pending patent applications, and any and all divisions, continuations, continuations in part,

reissues, continuing patent applications, reexaminations, and extensions thereof, any counterparts claiming priority therefrom, utility models, patents of importation/confirmation, certificates of invention, certificates of registration and like rights; (iv) works of authorship, copyrights and all other copyrightable works; (v) customer or end user lists, contact information, licensing and purchasing histories, manufacturing information, business plans and product roadmaps; (vi) technology, computer programs, computer software, including source and object code, application programming, firmware, user interfaces, manuals, models, firmware, algorithms and implementations thereof, development tools, flow charts, programmers' annotations and notes, and other work product used to design, plan, organize, maintain, support or develop any of the foregoing, irrespective of the media on which it is recorded, and other software related specifications, materials and documentation; (vii) databases, data and data collections; (viii) Domain Names and uniform resource locators; (ix) Trade Secrets, other proprietary information, know-how, methodologies, processes, technical data, techniques, methods, compositions, ideas, procedures, concepts and tools (whether or not patentable or reduced to practice), formulas, business and technical information, know-how, non-public information, and confidential information and rights to limit the use or disclosure thereof by any Person; (x) product designs, reference designs, specifications and documentation and (xi) moral rights. With respect to each of the foregoing, "Intellectual Property" includes all: (a) claims, causes of action and defenses relating to the enforcement of any of the foregoing, including for past infringement, (b) the goodwill associated with any of the foregoing; and (c) all tangible documentation relating to any of the foregoing including registrations of, applications for the registration of, and renewals and extensions of any of the foregoing with or by any Governmental Authority.

"<u>Intellectual Property Contracts</u>" means the Outbound Intellectual Property Contracts and the Inbound Intellectual Property Contracts.

"Knowledge" for purposes of this Agreement:

(a) An individual will be deemed to have Knowledge of a particular fact or other matter if that individual is actually aware of the fact or matter after due inquiry.

(b) A party, other than an individual, will be deemed to have Knowledge of a particular fact or other matter if any individual who is serving as a director, manager or officer of that party (or in any similar capacity) has Knowledge of that fact or other matter (as set forth in (a) above).

"<u>Law</u>" means any statute, rule, regulation, code, ordinance, resolution, order, writ, injunction, judgment, decree, ruling, promulgation, policy, treaty, directive, interpretation or guideline adopted or issued by any Governmental Authority.

"<u>Liability</u>" means with respect to any Person, any liability or obligation of such Person of any kind, character or description, whether known or unknown, absolute or contingent, accrued or unaccrued, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, due or to become due, vested or unvested, executory, determined, determinable or

otherwise, and whether or not the same is required to be accrued on the financial statements of such Person.

"<u>Licensed Company Intellectual Property</u>" means any Intellectual Property that is used Company, which is licensed to any Company by any Person.

"<u>Licenses</u>" means all licenses, franchises, accreditations and registrations, permits, approvals and consents (and all applications therefore) issued by any Governmental Authority in connection with the ownership, operation or development of any portion of the Business.

"<u>Lien</u>" means any mortgage, pledge, assessment, security interest, lien, adverse claim, levy, charge or other encumbrance of any kind, or any right of first refusal, conditional sale contract, option, title retention contract, or other contract to give or to refrain from giving any of the foregoing.

"<u>Material Adverse Effect</u>" means any event, change, effect or circumstance that has occurred that would reasonably be expected to have, individually or in the aggregate, a material adverse effect upon the Business, assets, Liabilities, financial condition, or operating results of one or both of the Companies; <u>provided</u>, <u>however</u>, that for purposes of this Agreement, a Material Adverse Effect shall not include the effect of (i) changes to the industry or markets in which the Business operates, (ii) the announcement or disclosure of the transactions contemplated herein or the identity of Buyer or its affiliates, (iii) general economic, regulatory or political conditions or changes, (iv) changes in the condition of financial, banking or securities markets (including any disruption thereof and any decline in the price of any security or any market index), (v) military action or any act of terrorism, and (vi) changes in Law after the date hereof; <u>except</u> (other than in the case of clause (ii) above), to the extent such effect, development, occurrence, circumstance, state of facts or change that has a materially disproportionate and adverse impact on any Company or the Business, taken as a whole, relative to other participants in the industries in which any Company conducts its business.

"Material Contracts" has the meaning set forth in Section 4.10.

"Material Customer" has the meaning set forth in Section 4.21(c).

"<u>Materials of Environmental Concern</u>" means chemicals, pollutants, contaminants, hazardous materials, hazardous substances and hazardous wastes, medical waste, toxic substances, petroleum and petroleum products and by-products, asbestoscontaining materials, PCBs, and any other chemicals, pollutants, substances or wastes, in each case so defined, identified, or regulated under any Environmental Law.

"Measurement Period" means the 12-month period beginning on the Closing Date.

"<u>Net Working Capital</u>" means the difference between (a) Current Assets of the Companies as of the close of business on the Closing Date, and (b) Current Liabilities of the Companies as of the close of business on the Closing Date, determined in accordance with

GAAP (except with respect to the Estimated Closing Statement as contemplated by <u>Section 2.3(a)</u>).

"Offer Letters" has the meaning set forth in Section 2.5(d).

"<u>Open Source License</u>" means any license meeting the Open Source Definition (as promulgated by the Open Source Initiative) or the Free Software Definition (as promulgated by the Free Software Foundation). For the avoidance of doubt, Open Source Licenses include Copyleft Licenses.

"<u>Open Source Materials</u>" means any software or other Intellectual Property subject to an Open Source License.

"<u>Options</u>" means stock options in i3V issued pursuant to, and subject in all cases to, the i3V 2018 Equity Incentive Plan as may be amended from time to time.

"Order" means a judgment, order, writ, injunction, decree, determination, or award of any Governmental Authority.

"<u>Owned Company Intellectual Property</u>" means any Intellectual Property that is used by any Company other than Licensed Company Intellectual Property.

"Parties" has the meaning set forth in the Preamble.

"<u>Payment Network</u>" means Mastercard International, Inc., Visa International, Inc., Visa USA, Inc., DFS Services LLC, American Express Travel Related Services, Inc., any affiliate of any of the foregoing and any other card association, debit card network or similar entity with whom any Company has a direct or indirect merchant or sponsorship relationship.

"<u>PCI-DSS</u>" has the meaning set forth in <u>Section 4.25(b)</u>.

"Percentage Interest" means 79.0% with respect to Graves and 21.0% with respect to Humphries.

"<u>Permitted Encumbrances</u>" means (a) any statutory Liens for current Taxes, assessments or other governmental charges not yet due and payable; (b) mechanics', carriers', workers', repairers' and similar Liens arising or incurred in the ordinary course of business that are not yet due and payable and which are not, individually or in the aggregate, material to the business, operations and financial condition of the property leased by the Companies so encumbered, taken as a whole; (c) zoning, entitlement, building and other land use regulations imposed by governmental agencies having jurisdiction over any property leased by the Companies which are not violated by the current use and operation of such property; and (d) covenants, conditions, restrictions, easements and other similar matters of record affecting title to any of the property leased by the Companies, which are or would be disclosed in policies of title insurance which, individually or in the aggregate, do not materially impair the occupancy or use of such property for the purposes for which it is currently used or proposed to be used in connection with the relevant Person's business.

"<u>Person</u>" means any individual, corporation, company, body corporate, association, partnership, firm, joint venture, limited liability company, trust or governmental agency.

"<u>Personal Information</u>" means information that, alone or in combination with other information, relates to a specific, identifiable individual person, including individual names, social security numbers, telephone numbers, home addresses, driver's license numbers, account numbers, card information, cardholder data, email addresses, internet protocol (IP) addresses, and vehicle registration numbers.

"<u>Personal Property</u>" means all tangible and intangible personal property used or held for use in connection with the Business, including all equipment, furniture, fixtures, machinery, computers, appliances, telephones, switches, dialers, office furnishings, instruments, leasehold improvements, spare parts, all rights in all warranties of any manufacturer or vendor with respect thereto and rebates received in connection with inventory or any item described in this definition.

"<u>Platform</u>" means the assets, including any Personal Property, Intellectual Property, and Trade Secrets, used by the Business for: (a) providing payment processing services that enable End Users to authorize and settle payment transactions with Customers; (b) facilitating license applications or renewals between End Users and Customers; (c) providing credential or license verification; or (d) any software-based tool not described in items (a) – (c) through which any Company provides its products and/or services to Customers.

"Pre-Closing Tax Period" has the meaning set forth in Section 6.2(a).

"<u>Privacy Laws</u>" means all applicable Laws governing the receipt, collection, use, storage, processing, sharing, security, disclosure or transfer of Personal Information, including The Family Educational Rights and Privacy Act, The Children's Online Privacy Protection Act, the Communications Decency Act and the Payment Card Industry Data Security Standard.

"<u>Proceeding</u>" means any arbitration, audit, hearing, investigation, subpoena, litigation, suit or other similar action, whether or not by or before a Governmental Authority.

"<u>Processors</u>" means any processors, acquirers or sponsor banks, or originator depository financial institutions utilized by any Company or with whom any Company is otherwise certified to provide payment processing services associated with the Business, including Pace Payment Systems, Inc. and Chase Paymentech.

"Purchase Price" has the meaning set forth in Section 2.2.

"<u>QSub</u>" means a "qualified subchapter S subsidiary" as such term is defined in Section 1361(b)(3)(B) of the Code.

"<u>Real Property</u>" means all fee, leasehold and other interests in real property owned or leased by any Company, whether directly or indirectly, or otherwise used or held for use in connection with the Business, together with all buildings, improvements and fixtures and

construction in progress located thereupon and all appurtenances, rights of way and air, mineral or other rights related thereto.

"<u>Referral Source</u>" means: any Person, other than employees, officers or directors of any Company, that (a) markets, refers or resells Company Products or services to any Person (including Customers) or (b) is paid a portion of any compensation received in connection with origination of any Customer Agreement or sale of any Company Products or services.

"<u>Referral Source Agreement</u>" means any agreement between a Referral Source on the one hand and any Company on the other.

"Required Consents" has the meaning set forth in Section 4.3.

"<u>Representative</u>" has the meaning set forth in <u>Section 10.1</u>.

"<u>Rules</u>" means the bylaws, regulations and/or requirements that are promulgated by the Payment Networks, Processors, National Automated Clearing House Association (NACHA), Payment Card Industry security standards council or similar entities or organizations.

"<u>Seller Indemnified Parties</u>" means the Sellers, Owners and the agents, representatives, successors and assigns and affiliates of any of the foregoing.

"<u>Special Representations</u>" means those representations and warranties set forth in <u>Section 4.11</u>, <u>Section 4.15</u>, <u>Section 4.17</u>, <u>Section 4.23</u> and <u>Section 4.25</u>.

"Straddle Tax Period" has the meaning set forth in Section 6.2(a).

"<u>Survival Period</u>" has the meaning set forth in <u>Section 8.1</u>.

"Taxes" means: (a) any and all federal, state, local, foreign and other net income, gross income, gross receipts, capital gains, sales, use, ad valorem, unclaimed property, transfer, franchise, profits, license, lease, rent, service, service use, withholding, payroll, employment, excise, severance, privilege, stamp, occupation, premium, property, windfall profits, alternative minimum, estimated, social security, workers' compensation, unemployment compensation or insurance, customs, duties or other taxes, fees, assessments or charges of any kind whatsoever, together with any interest and any penalties, additions to tax or additional amounts with respect thereto; (b) any Liability for payment of amounts described in clause (a) as a result of transferee Liability or otherwise through operation of law; and (c) any Liability for the payment of amounts described in clauses (a) or (b) as a result of any tax sharing, tax indemnity or tax allocation agreement or any other express or implied agreement to indemnify any other Person.

"Tax Authority" means any federal, state or local taxing authority.

"Tax Proceeding" has the meaning set forth in Section 6.2(e).

"<u>Tax Return</u>" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any Schedule or attachment thereto, and including any amendment thereof.

"Third-Party PCI-DSS Component" has the meaning set forth in Section 4.25(c).

"<u>Trade Secrets</u>" means confidential and proprietary information, whether oral or written, including ideas, designs, concepts, compilations of information, methods, techniques, procedures, processes and other know-how, whether or not patentable, of any nature in any form, including all writings, memoranda, copies, reports, papers, surveys, analyses, drawings, letters, computer printouts, computer programs, computer applications, specifications, business methods, business processes, business techniques, business plans, data (including Customer and Referral Source data), graphs, charts, sound recordings or pictorial reproductions used by the Companies in the Business, all of which are protected by the doctrine of trade secret because of their confidential treatment.

"<u>Trademarks</u>" means unregistered and registered trademarks and service marks, trademark and service mark applications, common law trademarks and service marks, trade dress and logos, trade names, business names, corporate names, product names and other source or business identifiers, certification marks, slogans, brand names, assumed names, and all other indicia of origin used by any Company and the goodwill associated with any of the foregoing and any renewals and extensions of any of the foregoing.

"<u>Transaction Bonuses</u>" means any bonuses, success fees, severance payments, change of control payments, amounts earned, accrued, payable or otherwise existing as a Liability of the Companies under any long term incentive plan or similar arrangement, and any other amounts payable and unpaid at Closing (and not otherwise irrevocably waived or forfeited) to any Person by the Companies or Owners in connection with the Contemplated Transactions (including the Companies' portion of any payroll Taxes associated with such amounts).

"Transaction Documents" has the meaning set forth in Section 3.1.

"<u>Transaction Expenses</u>" means all fees, costs, charges, expenses and obligations unpaid at Closing that are incurred by any Company, any Seller or any Owner in contemplation of, in connection with or relating to the preparation for, and consummation of, the Contemplated Transactions, including (a) the preparation, negotiation and execution of this Agreement and all other agreements, certificates, instruments and documents delivered under the terms of this Agreement and the transactions contemplated hereby and thereby (including any due diligence review of the Companies); and (b) financial advisory and professional services provided by any of the Companies', Sellers' or Owners' bankers, counsel, brokers, consultants, accountants, advisors (financial or otherwise), agents and representatives to the extent related to the Contemplated Transactions.



i3 VERTICALS COMPLETES NET DATA ACQUISITION Updates Guidance for Fiscal 2019

NASHVILLE, Tenn. (April 8, 2019) – i3 Verticals, Inc. (Nasdaq: IIIV) ("i3 Verticals" or the "Company") today announced the completion of the acquisitions of Northeast Texas Data, LLC and Graves Humphries Stahl, LLC (collectively, "Net Data") for a combined purchase price of approximately \$24 million.

Greg Daily, CEO of i3 Verticals, commented, "We are excited to announce the completion of the Net Data acquisition, as we continue to expand our Public Sector vertical. We look forward to working with the talented and dedicated Net Data team of 70 employees built over 35 years. With the completion of this transaction, we have acquired new software to offer to our local government customers and we have expanded into an attractive Texas market."

Net Data provides software and services for municipalities and local governments, primarily within Texas. The software platform provides solutions for judicial management, property tax collection, court records, law enforcement and traffic violations, financial systems, document imaging and other programs critical to the day-to-day business of local governments.

"In our February 13, 2019, earnings release, we indicated that we had entered into three signed term sheets for potential acquisitions. On March 4, 2019, we announced the completion of two of those acquisitions. This acquisition is the third and final acquisition of those three deals. Including Net Data, our pro forma leverage ratio is currently approximately 2.5x, and we have approximately \$40 million in borrowing capacity under our current Credit Agreement. We are excited about our acquisition pipeline and our ability to expand i3 Verticals' reach within attractive vertical markets," concluded Daily.

Revised 2019 Outlook

Net Data is expected to produce its strongest financial results during the Company's first and second fiscal quarters ended December 31 and March 31, respectively. Net Data will begin to be consolidated into the Company's financial results at the beginning of the third fiscal quarter ending June 30, 2019. The Company is providing the following revised outlook for the fiscal year ending September 30, 2019, incorporating the projected financial impact of the Net Data acquisition announced today:

(in thousands, except per share amounts)		Outlook Range Fiscal year ending September 30, 2019		
Adjusted net revenue ⁽¹⁾ (non-GAAP)	\$	127,000 - \$	133,000	
Adjusted EBITDA (non-GAAP)	\$	36,500 - \$	39,500	
Adjusted diluted earnings per share ⁽²⁾ (non-GAAP)	\$	0.85 - \$	0.90	

 Under GAAP, companies must adjust, as necessary, beginning balances of acquired deferred revenue to fair value as part of acquisition accounting as defined by GAAP. For the 2019 outlook, the Company has removed the effect of these adjustments to acquisition date fair value from acquisitions that have closed as of the earnings release date.

2. Assumes an effective pro forma tax rate of 25.0% (non-GAAP).

With respect to the "2019 Outlook" above, adjusted net revenue, adjusted EBITDA and adjusted diluted earnings per share are non-GAAP financial measures. Reconciliation of adjusted net revenue, adjusted EBITDA and adjusted diluted earnings per share guidance to the closest corresponding GAAP measure on a forward-looking basis is not available without unreasonable

-MORE-

IIIV Completes Net Data Acquisition Page 2 April 8, 2019

efforts. This inability results from the inherent difficulty in forecasting generally and quantifying certain projected amounts that are necessary for such reconciliations. In particular, sufficient information is not available to calculate certain adjustments required for such reconciliations, including changes in the fair value of contingent consideration, income tax expense of i3 Verticals, Inc. and equity-based compensation expense. The Company expects these adjustments may have a potentially significant impact on future GAAP financial results.

About i3 Verticals

Helping drive the convergence of software and payments, i3 Verticals delivers seamlessly integrated payment and software solutions to small- and medium-sized businesses and other organizations in strategic vertical markets, such as education, non-profit, the public sector, property management, and healthcare and to the business-to-business payments market. With a broad suite of payment and software solutions that address the specific needs of its clients in each strategic vertical market, i3 Verticals processed approximately \$11.7 billion in total payment volume for the 12 months ended December 31, 2018.

Forward-Looking Statements

This release contains forward-looking statements that are subject to risks and uncertainties. All statements other than statements of historical fact or relating to present facts or current conditions included in this release are forward-looking statements, including any statements regarding guidance and statements of a general economic or industry specific nature. Forward-looking statements give the Company's current expectations and projections relating to its financial condition, results of operations, guidance, plans, objectives, future performance and business. You can identify forward-looking statements by the fact that they do not relate strictly to historical or current facts. These statements may include words such as "anticipate," "estimate," "expect," "project," "outlook," "plan," "intend," "believe," "may," "will," "should," "could have," "exceed," "significantly," "likely" and other words and terms of similar meaning in connection with any discussion of the timing or nature of future operating or financial performance or other events.

-MORE-

The forward-looking statements contained in this release are based on assumptions that we have made in light of the Company's industry experience and its perceptions of historical trends, current conditions, expected future developments and other factors we believe are appropriate under the circumstances. As you review and consider information presented herein, you should understand that these statements are not guarantees of future performance or results. They depend upon future events and are subject to risks, uncertainties (many of which are beyond the Company's control) and assumptions. Although we believe that these forward-looking statements are based on reasonable assumptions, you should be aware that many factors could affect the Company's actual future performance or results and cause them to differ materially from those anticipated in the forward-looking statements. Certain of these factors and other risks are discussed in the Company's filings with the U.S. Securities and Exchange Commission (the "SEC") and include, but are not limited to: (i) the ability to generate revenues sufficient to maintain profitability and positive cash flow; (ii) competition in the Company's industry and the ability to compete effectively; (iii) the dependence on non-exclusive distribution partners to market the Company's products and services; (iv) the ability to keep pace with rapid developments and changes in the Company's industry and provide new products and services; (v) liability and reputation damage from unauthorized disclosure, destruction or modification of data or disruption of the Company's services; (vi) technical, operational and regulatory risks related to the Company's information technology systems and third-party providers' systems; (vii) reliance on third parties for significant services; (viii) exposure to economic conditions and political risks affecting consumer and commercial spending, including the use of credit cards; (ix) the ability to increase the Company's existing vertical markets, expand into new vertical markets and execute the Company's growth strategy; (x) the ability to successfully complete acquisitions and effectively integrate those acquisitions into the Company's services: (xi) degradation of the quality of the Company's products. services and support; (xii) the ability to retain clients, many of which are small- and medium-sized businesses, which can be difficult and costly to retain; (xiii) the Company's ability to successfully manage its intellectual property; (xiv) the ability to attract, recruit, retain and develop key personnel and qualified employees; (xv) risks related to laws, regulations and industry standards; (xvi) the Company's indebtedness and potential increases in its indebtedness; (xvii) operating and financial restrictions imposed by the Company's senior secured credit facility; and (xviii) the risk factors included in the Company's Annual Report on Form 10-K for the year ended September 30, 2018. Should one or more of these risks or uncertainties materialize, or should any of these assumptions prove incorrect, the Company's actual results may vary in material respects from those projected in these forwardlooking statements.

Any forward-looking statement made by us in this release speaks only as of the date of this release. Factors or events that could cause the Company's actual results to differ may emerge from time to time, and it is not possible for us to predict all of them. The Company undertakes no obligation to publicly update any forward-looking statement, whether as a result of new information, future developments or otherwise, except as may be required by law.

Contacts:

Clay Whitson Chief Financial Officer (615) 988-9890 cwhitson@i3verticals.com Scott Meriwether Senior Vice President - Finance (615) 942-6175 smeriwether@i3verticals.com

-END-